

Received November 23, 2020, accepted December 8, 2020, date of publication December 23, 2020, date of current version January 5, 2021. Digital Object Identifier 10.1109/ACCESS.2020.3046770

A Survey of Energy and Spectrum Harvesting Technologies and Protocols for Next Generation Wireless Networks

ASHISH PADHY¹, SANDEEP JOSHI², (Senior Member, IEEE), SAINATH BITRAGUNTA², (Senior Member, IEEE),

VINAY CHAMOLA^{®2}, (Senior Member, IEEE), AND BIPLAB SIKDAR^{®3}, (Senior Member, IEEE)

¹Department of Electrical and Electronics Engineering, Birla Institute of Technology & Science (BITS) Pilani, Hyderabad Campus, Hyderabad 500078, India ²Department of Electrical and Electronics Engineering, Birla Institute of Technology & Science (BITS) Pilani, Pilani 333031, India ³Department of Electrical and Computer Engineering, National University of Singapore, Singapore 117583

Corresponding author: Biplab Sikdar (bsikdar@nus.edu.sg)

This work was supported by the Department of Science and Technology (DST)—Science and Engineering Research Board (SERB) under Project ECR/2018/001479.

ABSTRACT Energy harvesting (EH) and spectrum harvesting (SH) are two promising and useful green communication and networking mechanisms for the next-generation wireless networks. While the former techniques exploit ambient energy sources to scavenge energy, the latter exploit the unused or moderately used electromagnetic spectrum. With the advent of cyber-physical systems and the Internet-of-Things (IoT), the presence of tens of billions of low power sensor devices would soon be a reality. These small sensing devices would be present in many systems around us, such as home appliances, telecommunication devices, medical electronics, transport systems, etc. These miniaturized, low-power consuming devices may exploit EH and SH techniques for energy storage and communication. These EH-SH-enabled sensors or low-power nodes need to consume very little energy for sensing and communications. In this comprehensive survey paper, we first present the historical background of EH, and SH techniques, and their development over several decades. Specifically, we focus on EH-SH communication technologies and protocols for a wide range of systems and networks. We present a detailed survey of the various harvesting techniques and protocols from recent literature. Finally, we describe exciting open, intra-disciplinary, and inter-disciplinary challenges for further research on EH-SH communication technologies.

INDEX TERMS Cognitive radio, cooperative sensing, energy harvesting, Internet of Things, next generation wireless networks, spectrum harvesting.

I. INTRODUCTION

Energy efficiency [1]–[3] and spectrum efficiency [4], [5] are the two most important parameters for next-generation beyond-5G communication networks which have primary focus on green communications. These two parameters are also linked to the capacity and the power requirements of the communication network [6]. In conventional wireless networks, the energy and spectral efficiency issues have been mostly dealt with separately; there is a need to provide an integrated approach to the same. Further, with the advent of a multitude of services on wireless devices and

The associate editor coordinating the review of this manuscript and approving it for publication was Guangjie $\operatorname{Han}^{(D)}$.

the greater proliferation of these devices, there is an urgent need to have energy as well as spectral efficient solutions. The diverse energy requirements for various physical objects such as sensors are proliferating due to the advent of technologies like the Internet of things (IoT) and its variants, such as the Internet of medical things. These nodes need sufficient bandwidth and energy to communicate reliably and efficiently. There are various sensors, devices, and physical objects that are powered by batteries. However, in several remote applications, the frequent replacement of batteries is not a feasible option. Therefore, energy harvesting from ambient energy resources [7] is an excellent alternative to enhance node lifetime, and hence, the network lifetime.

In recent years, the Internet has proliferated, connecting humans and systems worldwide so that uniquely recognizable devices can interact wirelessly with each other, supporting various kinds of applications and people on a wide scale. Thus, the conventional Internet has evolved into another intelligent network infrastructure called the IoT [8], [9]. A recent survey by Cisco [10], projected a 2.4-fold increase in machine-to-machine (M2M) connections from 6.1 billion in 2018 to 14.7 billion by 2023, and 5.3 billion users, which is 66% of the global population, will be connected to the Internet by 2023. IoT's highlights include hard and fast observation, dependable transmission, and smart functioning; "hard and fast observation" alludes to gathering and obtaining data anywhere and anytime by utilizing the recognition and estimation strategy, "dependable transmission" implies to have devices connected through a data network, such that accurate information is interchanged and transmitted by combining communication networks, and "smart functioning" corresponds to investigating the broad set of information and data from various industry sectors and businesses through different smart technology applications, such as big data, machine learning, etc., to improve visibility into the practical world and economic and social activities to realize smart decision-making and measures.

As IoT continues to grow rapidly, it is becoming a massive network that links almost every element of our activities [11]. IoT devices are usually powered using batteries. This results in high maintenance costs, with the replacement of batteries being difficult and expensive. Additionally, in certain environments such as industrial sensor nodes, sensors are installed in areas that are challenging to access or unreachable. Therefore, energy efficiency is critical in these applications. Further, for IoT devices, optimal energy storage and management frameworks [12], [13] are designed to realize the long and routine functioning of devices. The advent of cognitive radio (CR) technology has been perceived as a productive strategy to tackle spectrum shortages and low utilization scenarios using optimum sensing techniques. Security is also an important aspect to be considered in IoT [14].

Energy sources are of two types: non-renewable and renewable [15]. Examples of the former are nuclear, oil, coal, and natural gas, which are limited and would eventually be exhausted. On the other hand, the latter type of resources is readily available or replaceable within a short time, such as solar, wind, and biomass. Note that these renewable energy sources, though environment friendly, are not very efficient, and their performance depends on several factors. However, green research communities believe that green or renewable energy resources are useful for several next-generation green communication systems and networks. Techniques that use these renewable energy resources to harvest energy are called energy harvesting techniques [16], [17]. As discussed earlier, energy harvesting provides an opportunity with unique features for IoT and next generation wireless communications that conventional battery operated devices cannot provide. According to [18], the energy harvesting market amounts up to USD 440.39 million in 2019, which will increase to USD 817.2 million in 2025. Further, according to [19], the cognitive radio market is expected to reach USD 8.31 billion by 2024. In the electromagnetic spectrum, the useful frequencies suitable for communications are limited and licensed, such as 5G radio-frequency (RF) spectrum, TV frequency bands, and RF bands. However, most of the electromagnetic spectrum is underutilized. Therefore, for efficient utilization of the precious EM spectrum, the cognitive devices or radios need to be smart enough to sense and acquire the unused spectrum accurately. Spectrum harvesting (SH) [20] involves techniques that make the scarce spectrum resources available for various applications.

The motivation behind using EH and SH in next-generation communication networks, in general, can be summarized as follows:

- Achieving energy efficiency: In next-generation communication networks, one of the main concerns is to increase the battery life, which powers the wireless devices. Cognitive radio (CR) [21]–[24] assisted energy harvesting techniques are discussed in the literature, which can be combined with different energy harvesting techniques to provide energy-efficient solutions. The energy harvesting can be done from natural sources like wind and solar energy or radio frequency sources, also called RF energy harvesting. The important consideration is optimizing device operations, which may lead to extending the life of the devices on battery.
- Achieving spectrum access: Spectrum access in next-generation communication networks can be classified as priority-based, i.e., based on the reliability and latency considerations, or as tier-based, i.e., based on the location of wireless devices in the different tiers of the heterogeneous network [25]. The switching of wireless devices between different spectrum resources based on the device usage requirements is an important parameter.
- Achieving interference management: Next-generation heterogeneous networks are highly dense, and interference needs to be taken into account for reliable communication to take place. The inter-cell and the intra-cell interference are in the next generation networks like the device-to-device (D2D) communication networks [26], [27] make power management an important consideration along with interference management. Energy harvesting solutions using cognitive radio have been reported in literature and are helpful in reducing the interference.
- Achieving high data rate and low latency: The use cases envisioned for the next-generation communication networks are based on higher data rates and low latency, which are the typical characteristics of these communication networks. Efficient spectrum utilization is possible by using EH cognitive radio, which, along with the capability of energy harvesting providing higher energy, will help in achieving high data rates and low latency.



FIGURE 1. EH and SH in next generation wireless networks: A schematic flow showing the flow/ salient sections covered in the survey.

This paper presents a comprehensive review of the energy and spectrum harvesting technologies and protocols for next-generation wireless networks. The major contributions of this paper are summarized as follows:

- We present a detailed overview and comparison of the different EH techniques in next-generation wireless networks and discuss various EH techniques like the RF, ambient energy harvesting techniques.
- We provide an overview of SH techniques in next-generation wireless networks including CR based and cooperative sensing techniques and provide a detailed summary of the various narrowband and broadband sensing techniques.
- We give the future research areas and challenges for the EH and SH techniques along with the technology readiness level (TRL) description.

The remainder of the paper is organized as follows. Section II gives a review of the EH in next-generation wireless networks. Section III provides a review of the SH in next-generation wireless networks. Section IV discusses the hybrid systems and applications of EH and SH in these networks. Section V gives an overview of the future research areas, and finally, Section VI concludes the paper. Fig. 1 describes the brief outline of the paper.

EH has several appealing benefits and distinctive characteristics for next-generation wireless communications that are not offered by traditional battery or grid-power dependent communication systems. These include self-sustainable capacity, lower carbon footprint, and no requirement of battery replacement in wireless sensor nodes, with simple implementation in most of the complex scenarios. Thus, EH in wireless networks [28]-[33] is being studied extensively by the research community and being implemented in wide range of applications. Energy efficiency [34]-[36] and EH [37] are two ways to attain green communications [38], [39]. Different energy sources can be utilized, which can supply us with ambient renewable energy sources such as wind, solar [40], [41], motion, electromagnetic (EM) waves, water [42], [43]. In the past few years, there has been a lot of research on ambient EH [44]-[48]. The limitations of batteries such as leakages, limited storage capacity, charging, and discharging along with harvesting and management frameworks to optimize transmission are discussed in [49]–[52]. Designing the dimensions of photovoltaic (PV) panel and capacity of batteries for energy storage for solar-powered systems is discussed in [53]-[56]. Some studies have focused on radio frequency (RF) energy harvesting, which is capable of carrying both energy and information in wireless communication. A detailed study of RF energy harvesting using the ambient backscatter communication (AmBSC) system for passive IoT is done in [57]-[61]. Simultaneous wireless information and power transfer (SWIPT) is used to recover both information and energy from RF signals as detailed in [62]-[65]. Two architectures: power splitting [66]–[68] and time switching [69]-[71], are proposed which help to overcome the difficulties arising from practical circuit design constraints.

IEEEAccess

Scheduling of energy is also a primary method to manage the harvested energy to maintain *energy neutrality constraint* EH in wireless sensor networks; various *offline* [72]–[77] and *online* [78]–[80] methods have been discussed in the literature.

In [81], the authors presented information-theoretic limits and scheduling policies, and medium access control (MAC) protocols are presented in [82], [83]. In addition to it, the authors discuss simultaneous information and power transfer. Further, the authors state that the challenges lie in various layers as well as in the integration with the development of circuits and systems that harvest energy and transfer information. In [84], the authors focus on RF-EH and summarize advances in EH wireless communications and networks. The authors cover a wide range of topics viz., physical layer information-theoretic limits, and MAC protocols. The authors emphasize the development of efficient protocols for EH networks with energy and information transfer capabilities. Another survey paper on EH in IoTs [85] gives a classification of approaches to optimize harvested energy, their comparisons, analysis along with the proposed solutions. Furthermore, it also presents standards for interoperability.

Sharing of spectrum in the context of cognitive radio (CR) is discussed in [86], [87], and spectrum assignment in cognitive radio based IoT is studied in [88]. Routing mechanisms for cognitive sensor networks are proposed in [89]. A collaborative spectrum sensing mechanism with energy shortfalls is proposed in [90]. Spectrum sensing in CR is discussed in [91]-[93] where the radio senses the spectrum and transmits information simultaneously in the whole time slot. Opportunistic spectrum access with modelling using Markov chains is presented in [94]. A collaborative CR to schedule for energy efficiency and increased throughput for secondary users is framed in [95]. A MAC protocol for use in spectrum sensing is demonstrated in [96]. In existing literature, narrowband spectrum sensing [97]-[99] and wideband spectrum sensing [100], [101] techniques have been extensively described for CR. Narrowband spectrum sensing is used to verify the presence of primary users and is further categorized into various techniques such as energy detection [102]-[105], matched filter detection [106]-[108], feature detection [109]–[112], and eigenvalue based detection [113], [114]. Similarly, wideband spectrum sensing is used to detect the available spectrum holes which is classified based on Nyquist and sub-Nyquist sampling [115], multiband sensing [116]–[119], wavelet-based sensing [120]–[122], filter-bank based sensing [123]–[127], compressed sensing [128]–[135], and multi-coset sensing [136]-[139].

Cooperative spectrum sensing [140], [141] enhances sensing performance by using the spatial diversity feature of wireless communications [142]. Fusion algorithms are proposed for centralized topology based cooperative spectrum sensing (CSS) using directional antennas in [143]. A reliabilitybased cooperative decision fusion [144], and integration of belief propagation with fast detection framework [145] for CSS has also been proposed in the literature. A reputation based CSS that is robust against attacks is discussed in [146]. Spectrum opportunities are found using clustering in heterogeneous CR networks in [147]. To suppress the malicious interference in CR networks, various frameworks are modelled in [148], [149]. Cluster based cooperative spectrum sensing has been studied by the authors in [150]–[152].

Green powered cognitive networks or green networks [153]–[159] have motivated research on energy efficiency and EH to capture and store ambient energy. In [160], the authors provide a summary of energy management techniques in IoT. In [161]-[163], scheduling for EH and spectrum sensing based heterogeneous CR sensor networks is discussed in the context of green communications. Physical layer security is also a primary concern in EH based CR, which is studied in [164]. Time scheduling for backscatter-aided RF powered CR networks is discussed in [165]. To overcome the shortcomings due to spectrum and battery resource allocation, EH-based cognitive M2M communication is proposed in [166]. Resource allocation in wideband sensing-based CR considering various constraints are provided in [167]-[169]. CR technology offers the ability to maximize spectrum usage as much as possible under the resource assignment and EH schemes. In [170], the authors present a technique where the CR does EH from RF signal. Resource allocation in wideband CR with SWIPT is studied in [171]. Optimal resource management and allocation for CR based sensor networks with EH are proposed in [172]. EH from the ambient environment is studied in [173], where slot mode based CR is proposed to increase throughput. A hybrid CR having sleep mode is considered in [174] to harvest energy from the primary users. In [175], a dynamic spectrum sensing mechanism with energy efficiency using a power assignment framework is proposed. A novel non-orthogonal multiple access (NOMA) assisted overlay spectrum sharing for multi-user CR networks for improved spectrum utilization with two scheduling schemes is proposed in [176].

Relay assisted CR for EH is proposed in [177], [178]. Interference-aided energy harvesting with decode-andforward relaying (DF) in CR is studied in [179]. SWIPT based EH and DF relay-based cooperative spectrum sharing is proposed in [180]. In [181], [182], CR with EH is proposed. In [183], [184], the secondary and primary users cooperate for EH and to maximize the throughput. Cooperative spectrum sensing with RF energy harvesting employed for a multi-channel CR is proposed in [185]. A collaborative spectrum sensing with clustering of CR nodes is studied in [186], where spectrum sensing, EH, and information transmission is studied in a time slot. Relay based cooperative spectrum sensing is studied in [187]-[189] where RF and SWIPT-based EH schemes have been proposed. Cooperative sensing using clusters is proposed by the authors in [190], [191] for spectrum sensing and energy efficiency. In [192], cognitive wireless powered communication networks have been proposed for both underlay and overlay CR for wireless energy and information transfer.

II. ENERGY HARVESTING IN NEXT GENERATION WIRELESS NETWORKS

The interest in green communication has prompted a tremendous amount of research in related areas due to the significant rise in energy utilization by the communication and information sectors [39]. Green communication may be attained from two primary methods. The first one is consuming less energy with more information transmission, classified as energy efficiency [34], [35]. The second is by extracting and then transforming the energy from ambient environments, such as solar energy, wind energy, thermal power, etc., as shown in Figure 2, which is termed as energy harvesting [37], [2]. EH, also known as energy scavenging, focuses on obtaining clean energy from ambient renewable energy sources. Historically, researchers investigated harvesting energy from windmills, waterwheels etc. [42]. Figure 3 shows the 'power' waves of innovation [43]. In it, renewable energy is shown as the sixth power wave. Among its seventeen goals, climate action is one of the sustainable development goals (SDGs) listed by the United Nations (UN). The generation, transmission, and distribution of green and clean energy could alleviate the carbon footprint problem and contribute to achieving the critical SDG. Given its eco-friendly nature and other advantages such as improved network lifetime, researchers, industry, and academia worldwide are working on novel energy harvesting technologies and their integration with various systems and networks.



FIGURE 2. Popular renewable energy resources for scavenging clean and green energies.

For several decades, researchers and EH enthusiasts have explored EH methods, such as thermal and vibration. There are two motivating forces behind the quest for novel EH techniques: I) Desire to design autonomously powered sensor networks or mobile devices without batteries; II) Motivation to



FIGURE 3. 'Power waves' of innovation [43].

address climate change or global warming issues due to CO_2 emissions. Figure 2 shows some popular renewable energy sources to scavenge energy via EH techniques. Each of these EH technologies has some merits and some limitations in terms of place and environment conditions. Furthermore, these EH technologies can find their place in big cellular communication network entities (e.g., green gNodeBs) and also in micro-scale low power nodes, for example, IoT sensors in cyber-physical systems.

It is possible to integrate two or more of these renewable energy resources to harvest more energy [40] at a larger scale, depending on the application and requirements. For instance, a cellular network could use a hybrid solar and wind base station (BS) [41], [53] to serve its subscribers or mobile users. If the energy harvested by a BS is below a certain threshold, the BS could switch to an alternate non-renewable energy source. Similarly, it is also possible to exploit other renewable energy sources for powering various physical objects or sensors in cyber-physical systems.

The research community has extensively published papers focusing on different techniques and applications of energy harvesting. Many recent green technology developments have attracted researchers to recognize a new change in the power supply framework due to fossil fuel usage in wireless network infrastructure. If the IoT sensors are deployed in a remote area for monitoring of the environment (e.g., forests), frequent replacement of batteries is infeasible. EH technologies are entirely appropriate for these applications. The low power sensors could harvest energy from ambient sources such as solar and wind. Some of these applications are shown in Figure 4. Not all EH technologies are appropriate for these heterogeneous systems and networks. For instance, wind-based EH is reasonable to power up a road-side vehicular infrastructure where air-flow is abundant. On the other hand, RF-based EH is suitable for low power IoT devices. Therefore, the green EH communication system designer should apply these EH techniques optimally.

A. AMBIENT ENERGY HARVESTING

Conventional communication systems that function using an abundant energy-efficient cell and those communication systems for which energy from environmental resources is



FIGURE 4. EH wheel consisting of various technologies and applications.

accessible to communication nodes for energy harvesting, are irregular and vary with time. Energy-efficient base stations in green cellular networks where energy is drawn from the grid is also considered in [3]. However, an unlimited quantity of energy is theoretically available, by using the wireless sensor system's energy-neutral operation, the node's energy consumption is always lower or equal to the energy obtained from the environment, so energy is stored in storage devices of limited capacity, leading to energy overflow or leakage. Figure 5 shows a wireless power scavenger consisting of a receiver antenna, a matching network, a radio frequency to direct current (RF to DC) converter or rectifier, a unit to manage power, and store energy. After complete charging of battery power is supplied to sensors. Generally, transmission power is a specific cause of energy usage; however, in some situations, alternative means of energy usage at the transmitter can influence radiated power. These design concerns, particularly energy considerations, are circuit processing power, battery leaks, and charging/discharging faulty batteries' ineffectiveness. In [54] for a base station, tolerable outage probability with constraints on photo-voltaic (PV) panel size and number of batteries is proposed. Similarly, in [55], necessary battery capacity, and PV panel sizing based on resource availability and detailed design of solar powered base stations [56] is formulated. Maximizing transmission rate and offline throughput using adaptable rate strategies with various energy impacts on costs such as currents due to battery leakage and storage shortfalls are considered in [49], [50]. The effect of the incompetence of battery charging / discharging upon the optimum transmission power strategy with throughput is examined in [51]. A framework for models on single-user and distribution channels is found



FIGURE 5. Block diagram of a typical power scavenging module powering a communication transceiver [65].

to collect, extract, and use the battery's harvested energy. In [52], throughput efficient transmission mechanisms that consider non-ideal circuit power are considered for wireless energy harvesting with unlimited battery storage space for energy. A two-phase transmission framework is followed in offline strategies, where in the first step, the optimum transmission is on-off, the continuous transmission is best during the second step. Ultimately, a closed-loop online algorithm is suggested based on an offline solution.

In general, three protocols for ambient energy harvesting and usage have been suggested [44]-[47], namely: 1) Harvest-Use (HS): The correspondence node is driven directly by systems for extracting resources with no buffer for storage, 2) Harvest-Store-Use (HSU): Here, the harvested energy is used only after storage and 3)Harvest-Use-Store (HUS): It is instantly possible to use extracted energy which is briefly contained in a super-capacitor, and residual energy after processing is completed is stored; The HUS and HSU schemes are less efficient than the HU scheme. A harvestthen-cooperate (HTC) protocol is proposed in [48] where the downlink source and relay harvest energy from the access points, during collaborative work in the uplink for information transmission from the source. However, EH using naturally occurring sources gas not been as productive as they were anticipated since ambient sources are abnormal and unpredictable. There are however, specific environments and circumstances where they are relevant and appropriate.

B. RF ENERGY HARVESTING

Recently, some studies have suggested energy harvesting in wireless networks via RF signals. A distinct feature of RF energy harvesting is that RF signal carries both information and energy, which enables energy harvesting nodes to forage energy and receive information. RF energy harvesting is also suitable for long-distance and mobile environments. Among the various EH technologies, we mainly focus on RF-EH for IoT. The choice of RF-EH-IoT in this survey is because, for low-power IoT sensors, RF-EH seems more promising and feasible. A passive IoT device in a symbiotic radio system is referred to as the backscatter device (BSD). The most popular paradigm for the symbiotic radio is the ambient backscatter communication (AmBSC) system. AmBSC system uses ambient energy harvesting BSDs that harvest energy using ambient RF signals such as TV, Wi-Fi (ISM bands), FM, and RF radiated from other ambient sources. These BSDs backscatter the information signals by switching the impedance of the antenna. Two illustrations of AmBSC systems with single and multi-antenna tags are shown in Fig. 6, and Fig. 7, respectively.



FIGURE 6. An illustration of the AmBSC system having a single antenna ambient EH tag, with Tx denoting the primary transmitter, and Rx representing the primary receiver with the backscatter link via tag.





AmBSC system with multiple tags: In a large-scale passive IoT system, there could be many tags equipped with a single antenna or multiple antennas. Partial or best tag selection could be a practical solution that helps mitigate the problem of synchronization and RF chain complexity. However, the larger the number of tags, the higher the complexity of channel state information (CSI) and energy management. Specifically, acquiring accurate CSI is challenging because of limited pilot transmissions due to energy constraints. These issues lead to more technical challenges and open research problems in EH-SH passive IoT.

In [57], the authors proposed multi-antenna tag AmBSC where the backscatter system uses a K > 2 antenna-equipped tag; among these, K - M antennas are used for RF energy harvesting and M antennas for backscatter communication and the difficulty due blind detection of ambient backscatter signals from a multi-antenna tag via generalized like-lihood ratio test (GLRT) is also addressed. In [58], [60],

the authors considered a single semi-passive BSD. These tags (i.e., the BSDs) have constraints on energy, which is a precious resource and should be utilized and managed efficiently where signal-to-interference-plus-noise ratio (SINR) constraint for successive interference cancellation (SIC) is also considered for effective backscatter transmission. In [59], the authors proposed a spectrum sharing AmBSC system. In the works mentioned above and the references therein, authors analyzed PHY performance measures such as channel capacity and bit error rate, and also addressed detection and resource allocation problems.

Table 1 shows a comparison of various studies on AmBSC systems in passive IoT. There exists a fair amount of literature on EH-tags assisted backscatter communication in passive (or semi-passive) IoT. The analytical results in the literature on AmBSC include detection, channel estimation, performance analysis, channel coding, EH, and transmission mechanisms (e.g., wireless power transfer) or protocols, to name a few. However, there are still several open problems in PHY, MAC and layers above. For instance, these include space-time block coding, non-orthogonal multiple access, joint spectrum and energy management, optimization, etc.

Remarks on AmBSC-enabled passive IoT: Single antenna versus multiple antenna tags: Unlike a single EH and transmitting antenna tag, a multi-antenna tag can simultaneously perform both EH and information transmission. However, this multi-antenna tag increases the AmBSC system complexity and cost. Furthermore, the issues of synchronization problems in multi-antenna tag-assisted AmBSC are higher. Antenna selection is a useful solution to address these timing issues. However, for the tag-antenna selection, partial or full CSI is essential. Thus, there is a tradeoff between AmBSC system complexity and performance.

C. SIMULTANEOUS WIRELESS INFORMATION AND POWER TRANSFER

Receivers can recover both information and energy at the same time from wireless RF signals, described as SWIPT [63], and this approach has been suggested to distribute RF energy in sensor networks. SWIPT offers the benefit of providing configurable and reliable energy and information upon request. Nevertheless, it is unrealistic to perceive simultaneous energy harvesting and data transmission as entropy rate accounts for the amount of data. At the same time, RF signal's average squared value determines its power. Consequently, information transmitted and the energy harvested cannot be maximized simultaneously, which triggers requirements to redesign wireless networks with relevant constraints [64]. In reality, most of the methods are based on time switching or power splitting. In time switching architecture, time for information processing and RF energy harvesting is harmonized while power splitting focuses on the optimal division of received power into two parts: EH and information filtering.

In [66], a relay distributed *power splitting* is done using game theory for information forwarding and EH. In [67],

TABLE 1. Comparison of various AmBSC models.

Model	Constraints	Performance measures	Remarks
The multi-antenna-tag ambient backscatter system [57]	Energy constraints on tags	Probability of detection using GLRT	An ambient system composed of an RF source, a single-antenna reader and a multi-antenna tag (K >= 2), K-M antennas for energy harvesting, M antennas for back scattering.
The symbiotic radio system with a cooperative receiver [58]	Peak transmit power constraint, average transmit power constraint	Ergodic weighted sum rate	A symbiotic radio system with cooperative receiver consists of an active primary transmitter, a semi-passive BSD, and a cooperative receiver that receives and decodes both the PT's signal and the BSD's signal.
Opportunistic Ambient Backscatter Communication in RF-Powered Cognitive Radio Networks [61]	Interference and energy harvesting constraints	Energy efficiency in the presence of sensing errors	 AmBSC-harvest-then-transmit (HTT)- based cognitive radio network consists of a secondary user transceiver pair and a primary transceiver pair. The CR network in the opportunistic spec- trum access (OSA) paradigm, in which the PU channels are accessed opportunis- tically using SS to detect spectrum holes. The secondary is equipped with an energy-based spectrum sensing unit, an RF energy harvesting unit, and an AmBSC unit.
A spectrum sharing AmBSC communication system [59]	Average transmit power constraint, primary transmission outage constraint	Ergodic capacity of secondary system	The primary communication pair is a conventional communication system. The secondary communication pair is an ambient backscatter communication system, which consists of a wireless-powered passive tag and a battery-powered reader.
Cooperative ambient backscatter system [60]	Primary transmitter's Average transmit power constraint, SINR constraint for SIC	Average achievable rates	Three practical symbiotic transmission schemes, i.e., commensal, parasitic, and competitive schemes, were proposed for the cooperative ambient backscatter system

a novel, SWIPT based cooperative NOMA is discussed. The user with a strong channel operates as an EH relay by implementing power splitting and optimizing the beamforming vectors while improving QoS for the user with weak channel conditions. Both time-switching (TS) and power-splitting (PS) are employed in [68], where cooperative DF relaying is used for EH and information processing. Energy-limited relays do EH from received and co-channel interference signals; after that, the harvested energy is used for forwarding the decoded signal accurately to the destination. At high SNR, PS performs better than TS for achieving throughput from ergodic or outage capacity, while at low SNR, TS is superior to PS.

The authors [69] have proposed a dynamic *time switching* protocol where the switch between EH and data transmission is done using AF and DF relaying by smartly tracking the level of harvested energy at the relay. It is appealing as CSI is not needed, and relays do opportunistic use of harvested energy by charging their battery for transmitting the information. In [70], the authors describe a mechanism to balance the time duration between wireless power transfer and data transfer by employing an optimization to satisfy energy, time duration, and QoS constraints. In contrast to the sum usable

hput from superior efficiently it utilizes the harvested energy available [158], [193]–[195]. Unlike conventional battery-powered systems, controlling power in EH systems has to be in synchronization with energy consumption and the recharging of the battery because of the dynamic nature of energy availability in

harvested power.

OPTIMIZATION

tion with energy consumption and the recharging of the battery because of the dynamic nature of energy availability in nature. Energy outage and energy overflow create problems in harvested energy utilization. Energy scheduling techniques are of two types: *offline* and *online*, based on availability of knowledge of CSI and energy state information (ESI), causally and non-causally before transmission.

energy extracted from RF sources, the efficiency of commu-

nication of the EH method is defined using the harvested free

energy at disposal effectively. In [71], the authors consider the

non-linear features of EH circuits by employing a resource

allocation algorithm for the SWIPT system to maximize total

The performance of an EH system also depends on how

D. OFFLINE AND ONLINE ENERGY SCHEDULING AND

For *offline methods*, information about complete knowledge of CSI and ESI is known before transmission in the energy scheduling period. Reference [72] aims to minimize

packet loss rate by jointly optimizing energy allocation and scheduling algorithms based on channel, battery, and queue state information in industrial wireless sensor networks. In [73], the authors investigate the optimum allocation of power to minimize a non-convex average outage probability over transmit power in a fading channel. It is shown that the ideal power profile is non-decreasing over time and has a conserve-then-transmit framework, and a forwarding search algorithm that finds the global optimal offline power assignment is proposed with non-causal energy state information (ESI). However, an online power assignment strategy also has been proposed for causal ESI by considering dynamic programming and inspecting the system of optimal offline solutions. In [74], a linear combination of outage probabilities is minimized over a limited time period with predefined transmission rates. A piece-wise power assignment framework is developed for both cells with limited and unlimited storage capacities. A divide-and-conquer algorithm is suggested to find the optimal power allocation iteratively. Authors of [75] investigate the concave utility maximization optimization problem for EH-wireless sensor networks (EH-WSNs) where sum-rate maximization and distributed estimation related to water filling algorithm is presented. In [76], performance limitations of sensor nodes with finite energy where harvesting occurs variably is studied. A concave non-decreasing utility function is analyzed while managing the energy locally, achieving performance near to that with unlimited energy while keeping the probability of complete battery discharge low and addressing the problem of energy management in energy-replenishing nodes with finite battery and data buffers. In [77], a reinforcement learning model is proposed to allocate resources and schedule users in heterogeneous networks (HetNets) with hybrid energy supply to power small base stations.

Online strategies account for causal ESI and CSI as in [78], where a feasible rate maximization task is modeled as a Markov decision process (MDP) with continuous battery states having an access control structure of the EH transmitter that has maximum power limitations and works in a time-slotted fashion. The proposed algorithms produce an optimal continuous power allocation and better performance than discrete state MDP. Offline algorithms typically outperform online algorithms owing to the presence of non-causal information of energy arrivals and channels. For two instances where the harvested energy and channel can be causal and non-causal, offline and online power distribution structures are devised in [79]. The authors propose an optimal energy scheduling algorithm with an objective of sum-rate maximization of energy examined over limited time slots for k-user multi-access channels which do ambient EH. Due to the convex nature of the optimization problem, an iterative dynamic water-filling algorithm is suggested with limitations on cell capacity and energy utilization by transmitters. The authors maximize the networks' performance with the availability of EH sources in [80] with two reinforcement learning algorithms Q-learning and speedy Q-learning are used to formulate transmission methods in real-time scenarios by using joint randomness in arrivals of data and energy produced due to sensors and energy sources.

Table 2 shows a comparison among different energy harvesting algorithms discussed in this survey paper. Several MAC layer protocols have been proposed due to the differences in energy between nodes, adjusting to dynamic EH, and maintaining network efficiencies such as low latency and high packet delivery probability. Cross-layer scheduling structures have been developed between three layers: regulation of the source rate at the transport layer, optimizing multipath routing and flowrate at the network layer, and optimizing the MAC layer's duty cycle. In [83], a hybrid time division multiple access (TDMA) based medium access protocol has been suggested that avoids collisions and effectively uses the energy of the nodes and dynamically schedules the sleep/wake-up modes according to variability at the nodes. This increase the energy conservation by 40-60% compared to the IEEE 802.15.4-based MAC protocol.

E. MERITS AND DEMERITS OF EH TECHNOLOGIES

In the following subsection, we present various EH techniques and their challenges, merits, and demerits. Clean energy derived from renewable sources is good for the environment as the energy generation process does not contribute to the carbon footprint. With reliable and sustainable EH technologies, it is possible to confront this century's most challenging global warming and climate change problems. While EH technologies exhibit several merits, they also have demerits such as cost and complexity. The expenses such as CAPEX, OPEX, and EH system complexity vary from one EH technology to the other. A discussion on the exhaustive list of these features individually on each EH technology is beyond this survey paper's scope. However, we present a brief summary of these merits and demerits.

Merits:

- i) Renewable energy is good for the environment. Specifically, in wireless sensor networks, IoT, and cyber-physical systems, it is possible to significantly enhance network lifetime using these energy harvesting sensors or devices.
- ii) EH technologies help in waste reduction and management. For instance, it is possible to eliminate heavy batteries in complex subsystems.
- iii) Use of conventional non-renewable energy sources cause adverse effects on the environment. Furthermore, it is also difficult to power large numbers of IoT devices in the cyber cloud doing a wide range of sensing and communication tasks. For obvious reasons, EH technologies are an attractive option in these environments.
- iv) EH technologies are mostly reliable and even cost-effective in some applications such as wireless sensor networks.

Demerits: EH technologies do have demerits. Some of these demerits are listed below.

TABLE 2. Comparison between different energy harvesting techniques.

Harvesting Model	Design objective	System model	Advantages
Throughput-optimal joint selection and rate adaptation for multi-EH nodes [50]	Minimizing energy consumption	Stationary and Ergodic EH with fading channel	Channel and cell capacity does not affect transmission
EH from an environment with offline and online optimization [52]	Optimum transmit power allocation to maximize throughput	On-off transmitter with non-ideal circuit power in AWGN channel	Deals with the incompetence of battery charging / discharging
Harvest-the-cooperate model [48]	Optimum relay selection	AF relay with WEH in DL and WIT in UL in Rayleigh fading channel	Impact of relay position, relay number, time allocation considered
RF-enabled SWIPT with power splitting and time-switching [63]	Optimum transmission tradeoffs between maximum information rate vs energy transfer	MIMO broadcast system with two receivers	<i>Energy modulation</i> for maximum energy efficiency
Cooperative SWIPT-NOMA protocol employing EH and power splitting [67]	Maximizing data rate for a strong user and QoS requirement for weak user	MISO in AWGN channel	Relays driven by harvested energy and does not absorb energy from its battery
Dynamic time-switching [69]	Intelligent EH and information transfer	Energy-limited AF and DF relaying in quasi-static channel	No requirement of CSI, relay transmission power fixed and storage of energy at relay
Non-linear EH and resource allocation in SWIPT [71]	Non-convex optimization for maximizing total harvested power at EH receivers	Flat slow fading channel with downlink multiuser with beamforming	Practical effects of non-linearity in transmitter is considered
Time-switching and power-splitting with interference aided EH for cooperative DF relaying [68]	Ergodic and outage capacity with impacts of interference power and achievable throughput is derived	Rayleigh fading with Block-fading model with CSI available at receiver	Considers CCI which occur in wireless channels
MDP and online stochastic algorithm with multi-level water-filling framework for distributed energy assignment [79]	Extending lifetime of sensor nodes and minimize packet loss rate	AWGN with flat fading	Energy allocation and scheduling done with considering CSI, QSI, and BSI
Offline and online power allocation algorithms with EH limitations and CDI at transmitter [73]	Minimizing of outage probability	Block-fading channel with transmission over N EH periods and M communication blocks	Both causal and non-causal ESI
Optimal offline and online power control policy [74]	Minimizing the weighted sum of the outage probabilities	Rayleigh fading channel with finite and infinite cell capacity	Robust to prediction errors of the harvested energy
Offline and online scheduling algorithm using stochastic optimization problem employing a continuous MDP [78]	Maximizing achievable rate and optimal continuous power allocation	Slow fading channel with causal CSI and ESI with EH in time-slotted fashion and access-controlled transmitter	Better performance than conventional discrete MDP having finite- and infinite-horizon cases

- i) Several EH technologies are expensive; for example, in terms of initial deployment or infrastructure costs and materials cost.
- ii) EH technologies (mostly) occupy a significant amount of space. However, this varies drastically from one EH technology to the other.
- iii) Though most of the EH technologies are eco-friendly, some of them still cause some adverse effects on the environment, for example, biomass EH technology.
- iv) EH technologies suffer from low or medium energy efficiencies. There are several open problems and technical challenges associated with this demerit of EH technologies.

F. ENERGY EFFICIENT WIRELESS NETWORK CHARACTERISTICS

Wireless networks maybe classified as energy-efficient, or not, dependent upon the performance of the network and

the energy expenditure. In literature, studies on the energy efficiency of wireless networks can be classified based on the following definitions [32]:

• *bits per Joule:* Maximizing bits per Joule, or in other words,

$$\frac{\text{bits}}{\text{Joule}} = \frac{\text{capacity of the network}}{\text{throughput per unit of energy}}, \qquad (1)$$

covers both the performance of the network and the energy expenditure. The optimization of this parameter helps in classification into one of the categories under the energy-efficient wireless networks,

• *Quality of service*: Minimization of energy with guaranteed performance needs to take care of the Quality of Service from the network point of view and Quality of Experience from the user point of view. The energy minimization with the fixed Quality of Service or Experience is one of the optimization parameters which helps

in classification into one of the categories under the energy-efficient wireless networks.

• *Energy constraint*: In the cases with energy constraints such as battery-operated devices, networks with energy harvesting, and networks with limited energy, the energy-constrained optimization helps in classification into another category of energy-efficient wireless networks.

The next-generation networks are heterogeneous and need capabilities to process and acquire large amounts of data; this would pose major challenges for the network to remain energy efficient. Energy-efficient networks for handling data would need to take care of the aspects as described below [32]:

- Acquiring information: Acquisition of information that is relevant out of the large volumes of information available is an important consideration. The commonly used devices from which data can be sensed are sensors and mobile devices. For instance, in wireless sensor networks, the sensor arrays are used to collect data and are deployed, taking into consideration factors like the coverage and the energy efficient data collection methods. Techniques like those based on compressive sensing have been deployed in sensor networks for data acquisition [32] and are found to be efficient in energy terms and accuracy.
- Communicating the acquired information: The important consideration after data acquisition is the efficient communication of the information. In wireless devices, the primary source of energy consumption can be traced to power utilized during device operation, which is further dependent upon the time during which the device remains in the active state and resource allocation from the network. One possible solution for the offloading of the network traffic and making it spectrally efficient, as described in the literature, is the D2D communication technique. D2D communication has the advantage of providing spectral and energy-efficient communication as the users in proximity communicate directly with each other without the involvement of the network infrastructure.
- Storing the information efficiently: The storage of data requires energy-efficient and cost-efficient techniques, since the storage costs of data are high and the repeat access of the stored data also involves challenges. In wireless devices, one source of network traffic is downloading of popular files by different users from the network, thus leading to the download of multiple copies of the same file and congestion in the network. A solution to this problem is given by D2D communication where the users can directly communicate and share the file instead of downloading again from the network, leading to reduced congestion in the network; mobile caching is also one of the solutions for this problem.
- Computation of the information: To get analytical insights from the information which is acquired and

VOLUME 9, 2021

stored, efficient computation of the information is an important step. The energy consumption of a typical data center is around the same as that consumed by 25,000 households in [36], which shows massive energy consumption and points to the requirements for energy-efficient computations. In literature, it has been shown that computation offloading to cloud-based services is more efficient; by doing so, the computations are shifted from the mobile devices and save their power consumption and battery usage. In [17], the authors demonstrate the trade-offs between the costs involved in communication and computation while offloading to the cloud computing networks. It is shown that the energy that a node can save due to offloading can be expressed as

$$E_{save} = aA - \frac{b}{B}C, \qquad (2)$$

where *a* depends on the nodes power consumption, *b* depends on the server, and both are constants. *A* denotes the computation instructions, *C* denotes the bytes of information sent by the mobile device for offloading, and *B* denotes the bandwidth for communication. We can see from (2) that whether the energy is saved or not by offloading will depend on both *A* and *C* values.

G. EH: INFORMATION THEORETIC SENSE

Consider the received symbol in an additive white Gaussian noise (AWGN) channel to be

$$y = s + n, (3)$$

where *s* denotes the transmitted signal and *n* denotes AWGN with zero mean and unit variance. According to Shannon [81], the power constraint on the codewords for very large number of codewords *m* is given as

$$\frac{1}{m}\sum_{k=1}^{m}{s_k}^2 \le P\,,$$
(4)

where P denotes the power. At every channel usage, the causality constraint [81] that needs to be satisfied is given as

$$\sum_{k=1}^{n} s_k^2 < \sum_{k=1}^{n} E_k, n = 1, \dots, m,$$
(5)

where E_k denotes the stationary random ergodic process followed by the energy that is harvested at the transmitter such that $\mathbb{E}[E_k] = P$, with $\mathbb{E}[\cdot]$ being the expectation operator. This condition ensures that the codeword is transmitted without any outages in energy. Further, to save energy which is harvested for later applications, if battery is not available, then the constraint for the codewords is given as

$$s_k^2 < E_k, k = 1, \dots, m.$$
 (6)

If the battery is available with a finite size, with a maximum size denoted by B_{max} , then we can say that

$$E_{B_{k+1}} = \min\left(E_{B_k} - s_k^2 + E_k, B_{max}\right)$$
 (7)

which denotes that at first s_k^2 amount of energy is dissipated and then E_k amount of energy is harvested in the battery.

It has been shown in [81] that the capacity of the additive white Gaussian noise channel with the constraint as given by (5) is expressed as

$$C = \lim_{m \to \infty} \frac{1}{m} \max I(S^m; Y^m), \tag{8}$$

and further, the upper bound on the capacity can be expressed as

$$C \le \frac{1}{2}\log(1+P)$$
 . (9)

Energy capacity of the systems having nodes with buffer capability to store the energy harvested is larger, as shown in [16]. The channel capacity of an system under ideal conditions and having an infinite buffer is given as

$$C = \frac{1}{2} \log \left(\frac{\mathbb{E}[E_p]}{\sigma^2} \right), \qquad (10)$$

where E_p is stationary ergodic and denotes that the node is replenishing energy by E_p at time p and σ^2 denotes the noise variance.

III. SPECTRUM HARVESTING IN NEXT-GENERATION WIRELESS NETWORKS

The rising demand for wireless data is due to mobile devices unprecedented growth and Internet based applications. Due to the rapid increase in the number of IoT products, energy efficiency and spectrum efficiency are two critical concerns with IoT development. Machine-to-machine communications are perceived as one of the most likely innovations for IoT implementation in the 5G network. The 5G network [196]–[198] is flexible enough to allow a broad range of technologies and accommodate explosive traffic growth, thereby making headway for IoT development. Later on, 5G-based IoT can furnish a wide range of IoT systems with big data. As the evolution of the wireless communication systems continues, bandwidth requirements are also growing, and the increasing demand leaves wireless spectrum capacity even more inadequate. Cognitive radio has been recognized as a possible remedy for the scarcity of spectrum owing to its primary idea of dynamic spectrum access.

Figure 8 depicts the three phases of spectrum sensing, spectrum harvesting, and communicating tasks that a CR needs to accomplish for communicating opportunistically. Among these tasks, spectrum sensing is the unique signal processing task, which is essential for acquiring spectrum awareness for CRs in dynamic spectrum sharing networks. An overlay mode CR can acquire spectrum awareness via spectrum sensing (SS). SS is responsible for deciding on scanning spectrum spaces and is a unique characteristic of CR.

In the following subsections, we describe various spectrum sensing methods for next-generation wireless networks.



FIGURE 8. The phases of spectrum harvesting and communicating devices where the CR harvests minimum required bandwidth for communication.

A. SPECTRUM SENSING VIA CR

An intelligent wireless communication system, i.e., CR, could be a non-energy harvesting smart radio system or energy harvesting intelligent radio system. In the former, CR has to sense the unused EM spectrum and harvest bandwidth. On the other hand, in the latter, CR has to harvest both bandwidth and energy, which is more complex and highly limited in terms of its applications. For example, radio frequency (RF)-based EH-CR has to sense RF signals and harvest energy that can either be used on the fly or stored and then used for later tasks such as sensing and communication.

Figure 9 shows some of the applications and modes of CR. In the opportunistic mode, a CR, i.e., the secondary user (SU) or unlicensed user, could use the spectrum only when the primary or the licensed user is switched off. However, in underlay mode, both primary users (PU) and SU could co-exist. However, SU's transmissions should not cause interference at the primary receiver above certain thresholds. There are various technical and non-technical challenges in turning these applications and modes into reality. There do exist several tradeoffs among the operating modes of non-EH and EH CR and their performance. In the upcoming sections, we present various protocols developed in the literature on spectrum harvesting systems and networks. Later, we describe some exciting insights into EH and SH technologies from commercial and economic perspectives.

Figure 10 illustrates the motivation of spectrum sensing to find the spectral holes. A cluster-based wireless sensor networks (WSN) is also shown. Three clusters are present where sensor nodes send their information to the cluster head (CH), which is then sent to the sink node. In [86], the authors presented an overview of IoT-based cognitive radio systems and proposed that CR-based IoT systems are a feasible option for reliable and appropriate use of spectrum resources, including in the presence of primary users. They also provided a detailed survey on various architectures, frameworks, and spectrum-related functionalities of each layer in the CR-based IoT system. It is predicted that IoT without cognition can only be a strain on current infrastructure networks. The authors of [87] have reviewed a detailed







FIGURE 10. Illustration of spectrum sensing and cluster-based WSN [28].

study on the frameworks enabling sharing the spectrum in CR. Existing SS algorithms are also discussed, such as narrowband sensing, wideband sensing, and cooperative sensing, along with advances in various aspects of implementation such as power consumption, complexity, throughput, and performance. In [88], the optimum spectrum assignment concerning network throughput is examined, and the use of spectrum in IoT focused on CR. With mutual interference and resource allocation limitations under simultaneous transmission, a genetic algorithm optimization is proposed, with links acting as chromosomes, to achieve maximum spectrum utilization. Though the dynamic spectrum access mechanism helps the secondary users to access the wireless channels licensed to primary users due to high spectrum availability variations, it is difficult to establish an effective routing method for secondary users in cognitive sensor networks.

VOLUME 9, 2021

In [89], two cognitive sensor network (CSN) routing mechanisms are proposed, which account for spectrum channels' quantity and quality. One way to minimize rerouting is by re-transmission, considering the probability of successful delivery through all channels. In contrast, the other routing mechanism considers mean delay while transmission over all channels. Due to fading and shadowing, single secondary user performance is limited. For proper and effective wireless channel sensing, a collaborative spectrum sensing is proposed in [90], where the time slot is divided into two stages: spectrum sensing and reporting. The results reporting period depends on the number of collaborative sensing users, while the time invested in sensing the spectrum determines the number of sensing samples. Two scenarios with extra and shortfall of energy are considered, and a convex optimization problem is formulated for the overall probability of error with time and energy limitations. In [91], a full-duplex

cognitive radio network (FD-CRN) is used where throughput maximization for SU is done. FD-CRN can sense the spectrum and transmit data in the whole time slot. Two algorithms based on brute-force search and particle swarm optimization (PSO) methods to help the SU achieve optimal detection thresholds and maximize throughput is proposed in SU spectrum sensing and cooperative sensing environments. In [92], SS for multi-antenna CR using support vector machine (SVM) algorithms is proposed. The SS problem for multiple PUs is modeled as a multiple detection problem with a beamformer-aided feature realization method to boost the capacity of the SVM. Sensing of PUs spectrum occupancy state by SUs reduces the throughput for SUs. In [93], a simultaneous spectrum sensing and information transfer framework with imperfect signal cancellation and energy detection technique to detect PU is proposed. An opportunistic spectrum access (OSA) model is developed in [94], where SU performs OSA with a dynamic PU having a two-state random process modeled using Markov chains. The probabilities of detection and false alarm are established using chi-squared distributions and acts as test statistics for the energy detector.

A colony-based energy-efficient-sensor scheduling algorithm is proposed in [95] for a collaborative CR with constrained energy source using heterogeneous sensors, where optimal scheduling of sensors is performed in order to ensure the necessary output by sensing and increase the throughput of the secondary system. For effectively sensing of PUs, synchronization of secondary and primary networks is desired to decrease the probability of failure while transmission of PUs, but is challenging to achieve. In [96], a full-duplex MAC protocol SS framework for a multiple channel non-time slotted CR networks (CRNs) is proposed where SUs can sense PUs reactivation time in a timely manner while transmitting. The proposed protocol provides high throughput for PUs with efficient use of the channel by SUs.

As mentioned earlier, depending on the size of the band of concern, spectrum sensing approaches can be graded into two categories. Narrowband sensing [97] addresses the issue of determining whether a specific slice of spectrum is the hole. On the other hand, wideband spectrum sensing [100] is focused on categorizing individual wideband slices to be either occupied or empty. In reality, we require both the sensing technologies in the cognitive cycle. Usually, there exist two separate stages of PU detection in interweave CR. During the preliminary sensing process, in order to detect the accessible spectrum holes wideband spectrum sensing is used [101]. In compliance with specific requirements, the finest possible spectrum band is selected after identifying and examining spectrum holes. Nevertheless, due to high variations in mobile scenarios, the chosen narrowband may be occupied by the primary user once the communications begin. To overcome such difficulties, narrowband sensing by SU is done to verify whether PU is present or not prior to communication. Figure 11 shows various techniques as available in the literature that have been proposed to overcome the challenges in spectrum sensing.

B. NARROWBAND SPECTRUM SENSING

In interweave CR, before communication, SU must sense and verify whether spectrum is available or not, which is carried out by narrowband spectrum sensing. The authors of [98] suggest two architectures to perform narrowband spectrum sensing: single-radio and dual-radio. Merits of single-radio architecture are low power consumption, low cost, and simplicity, but it has less spectral efficiency [99]. Sensing by dual-radio architecture has demerits like increased power consumption and hardware cost. SU must be efficient in detecting the weak signals of PU so that it can rapidly evacuate the occupied spectrum.

We present below the various narrowband spectrum sensing techniques and recent developments as available in the literature. Further, techniques related to narrowband sensing are summarized with its classifications, its limitations, and associated references in Table 3.

Energy detection (ED) is a non-coherent detection technique where there is no requirement of prior information of PU signals and has less complexity. In [103], a kernelized ED scheme is proposed for various Gaussian and non-Gaussian impulsive noises, accounting for practical human-made noise conditions which comprise of higher-order and fractional lower-order moments using non-linear kernel functions for optimal sensing. As the ED method for spectrum sensing needs to be aware of noise power, it therefore has performance degradation due to noise uncertainty. To overcome this noise uncertainty and achieve reliable computing performance with low complexity, a blind ED for spatial spectrum sensing using maximum-minimum energy is proposed in [102], where an electronically steerable parasitic antenna receptor (ESPAR) is used, which can split the space into multiple sectors and switch the receive beam pattern to individual sectors in a time splitting approach. Matched Filter Detector (MF) uses coherent detection and maximizes the received signal-to-noise ratio (SNR) by applying cross-correlation between already known primary user signal and the received signal. Due to coherent detection, the SU needs to be completely aware of the characteristics of the PU signal, such as bandwidth, frequency of operation, the shape of the pulse, order, and format of modulation, etc. Due to its robust performance in low SNR conditions, MF detection is employed to detect weak signals. It has been shown that the efficiency and sensitivity of the MF detector decline quickly as mean noise power fluctuation increases and gets even poorer at low SNR conditions. This occurs due to a specified threshold; however, the performance can be enhanced by modifying this threshold. In [106], a dynamic sensing threshold is estimated to increase the efficiency of sensing detection. In [107], an MF detector is employed to recognize the presence of the primary user and its power level. This framework provides great flexibility to PU to vary its transmit power according to the situation of the environment. Even SU can modify its transmit





FIGURE 11. Classification of spectrum sensing techniques based on bandwidth.

power to fulfill the interference conditions for various power levels of PU. Authors of [108] have implemented a framework where SU has multiple antennas, and PU has a single antenna and the primary user's training signal set is known apriori by secondary users by cooperation. Even though MF detection has better performance, it requires knowledge about fading of the channel, frequency offset, or proper timing, or else the correlation becomes weak, which degrades the performance.

Feature detection comprises of a comprehensive class of spectrum sensing techniques. A subset of feature detectors, referred to as waveform detectors, is based on prior awareness of the construction of the PU signal. Waveform detection is coherent in nature, so it performs a correlation between the real known signal and the received signal. A comparison is then made between the output and the threshold for detecting the presence or absence of PU. In [109], first, a range decision test is proposed to set the threshold for comparison, which does not vary due to the variations in the SNR based on new decision criteria. In the second approach, SU senses the availability of channel while it is transmitting by using waveform detection so that SU's rate rises. Since PU signal undergoes modulation due to second-order periodicity, PUs have cyclostationary feature. These features are used for detecting the presence of primary users by PU signals as cyclostationary features are absent in stationary noise or ary features of the modulated signal, optimized recovery of generalized frequency division multiplexing (GFDM) signal is presented. By exploring the sparse property, the spectral correlation function (SCF) of the GFDM signal and the covariance function of its compressive samples are constructed. Further, a generalized likelihood ratio test (GLRT) is employed to find out the occupied spectrum. Numerical results prove that the proposed method has good performance at low SNR conditions. Detection of the GFDM signal has significantly better performance than detecting OFDM signals, so GFDM is a better technique that can be used in 5G and CR. Cyclostationary property is used in [111] to avoid the intercell interference in femtocell heterogeneous Long-Term Evolution Advanced (LTE-A) networks. It is shown that employing cyclostationary detection has better performance, particularly in the case of noise uncertainty, in terms of false alarms, detection probability, and deflection coefficient.

interference signal [97]. In [110], using the cyclostation-

Eigenvalue based detection: due to constraints of coherent detection and implementation of a practical system, various other methods have been proposed, one of them being eigenvalue based detection. In [113], [114] to reduce false alarm probabilities under noise uncertainty, a Max-Min eigenvalue detection is proposed where the ratio between maximum to minimum eigenvalues of the covariance matrix is set as a test statistic. This removes the need for PU and the transmission

Technique	Design model and References	Advantages	Remarks
Energy Detector (ED)	 Blind spatial SS [102]. Kernel ED in Gaussian and non-Gaussian impul- sive noises [103]. Adaptive algorithm for SS [104]. Four-level hypothesis blind detector for SS [105]. 	 Less complexity and operating cost with easy implementation. Non-coherent detection with no requirement of preliminary knowledge of primary network. 	 Performance degradation at low SNR conditions. Highly vulnerable to uncertainty in noise. Cannot differentiate between primary signal and noise. Unsuitable for spread spectrum methods.
Matched filter (MF)	 MF with dynamic threshold [106]. MF based SS and power level recognition using multiple antennas [108]. 	 Lesser sensing time and maximize the received SNR. Robustness against interference and stationary Gaussian noise. Better performance at low SNR. 	 Requires prior knowledge of PU. Dedicated synchronization sensing receiver for each SU. Large complexity and power consumption.
Feature detection (FD)	 Waveform detection by threshold comparison [109]. Quasi-cyclostationary feature (QCF) detector [112]. 	 Reliable sensing with detection at low SNR. Robust against uncertainty in noise. Distinction among types of transmission and primary sys- tems. 	 Slow sensing than ED and requires high cost and complexity. Prior knowledge of PU required. To achieve high accuracy a longer length of known sequence is needed thereby reducing efficiency of spectrum.
Eigenvalue-based detection (EVD)	 Throughput maximization of SU by eigenvalue- based SS [113]. Effect of noise correlation on eigenvalue SS [114]. 	 Robust against fluctuations in noise and detects signals with low SNR. Non-coherent detection and reliable sensing. 	• High computational complexity and time required for sensing.

TABLE 3. Summary of various narrowbanc	l sensing techniq	ues in	cognitive	radio n	etworks.
-----------------------------------------------	-------------------	--------	-----------	---------	----------

channel's prior information and overcomes noise uncertainty at the receiver.

C. WIDEBAND SPECTRUM SENSING

Sensing of bandwidth exceeding the channel's coherence bandwidth is done by wideband spectrum sensing, for example, ultra-high frequency (UHF) TV band (between 300 MHz and 3 GHz). With respect to the sampling frequency, two primary wideband spectrum sensing schemes have been proposed. The first approach is to sample the required spectrum using simple Nyquist frequency [115]. The second approach advocates the division of the sampling problem into multiple narrowband spectrum detection scenarios, or the edge between the occupied and vacant band is detected. These methods have large computational complexity and require high sampling rates, large sensing time, and processing delay. Algorithms related to wideband spectrum sensing with its classification, its brief description, and limitations are tabulated in Table 4.

Multiband Sensing or FFT-Based Sensing is proposed in [116], where sampling of a wideband signal is done by an analog to digital converter (ADC) at a high sampling rate. After dividing the samples into multiple segments, a Fast Fourier transform (FFT) is applied to estimate the

1752

power of each segment. Each segment is further divided into multiple narrowbands, and spectral occupancy is found out using binary hypotheses tests. Hardware implementation of FFT-based spectral correlation and pilot sensing SS framework is implemented in [117] on FPGA as per IEEE 802.22 standard. An adaptive multiband SS is derived in [118], where the framework incorporates two phases, the exploration phase and the detection phase, where several spectral holes are recognized among the surviving channels. A multiband SS technique and cross-layer reconfiguration resource allocation, i.e., the channel for IoT in cognitive 5G networks, is proposed in [119].

Wavelet-Based

Sensing is proposed in [120] for 60 GHz Millimetre-wave (mmWave) 5G small cell heterogeneous networks. The suggested framework can mitigate leakage in the spectrum and interference caused due to other prevailing networks in identical frequency bands by modifying the sub-carriers in accordance with cognitive information supplied by wavelet packet-based spectrum sensing (WPSS) and by utilizing a wavelet-based filter bank multicarrier (FBMC) modulation for reduction of sidelobes. Implementation of wavelet-based spectrum sensing is carried out in [121] for low SNR conditions. Wavelet transform (WT) is utilized as a tool that



	TABLE 4.	Summary	of various	wideband	sensing	techniques	in	cognitive	radio	networks.
--	----------	---------	------------	----------	---------	------------	----	-----------	-------	-----------

Technique	Design model and References	Advantages	Remarks
FFT-based Detector	• FFT algorithm used to identify multiple spectral holes [116]–[118].	 Nyquist sampling and Non-coherent detection. Medium complexity and up-grades acquisition performance. Removes the effects of phase rotation. 	• Requires high sampling rate and high power consumption.
Filter-bank Detector	• Prototype filters used with low sampling rate [125]– [127].	• Provides high performance and detects the dynamic nature WBSS.	High complexity due wide num- ber of RF front-ends devices.
Wavelet-based Detector	• PSD of wideband spec- trum used for exact lo- cation of spectral densi- ties [120], [122].	• Non-coherent and edge detec- tion with Nyquist sampling.	• High computational complexity.
Compressive Sensing	• Extract and detect the wideband signal directly to attain effective wideband sensing with a much lower sampling rate than the Nyquist scenario [129]–[135].	 Less power and sampling rate. Quick and precise detection of spectrum with sub-Nyquist sampling, non-coherent detec- tion. 	• High dynamic range, more sen- sitivity to noise and sparse na- ture.
Multi-coset Sensing	 Adaptive blind non- uniform Multi-coset samplers [136]. Discrete MC sampling [137]. Reconstruction of spec- trum [138], [139]. 	• Blind sensing, sub-Nyquist sampling, non-uniform quantizer.	• Need of synchronization cir- cuits.

produces information about the details on the precise frequency transition locations and spectral densities. Consequently, the wavelet transform is used to find out singularities in wideband power spectral density (PSD), where identification of the anomalies is considered an edge detection problem. In [122] an improved WT algorithm is derived in which non-linear logarithmic scaling of WT coefficients is carried out after normalization of the wideband PSD. In this framework, by carrying out such a process, WT-based edge detection achieves better accuracy and is more effective than conventional WT-based edge detection.

Filter-bank sensing: Fast convolution filter banks are studied in [123] for flexible and optimized processing of waveforms and sub-band energy detection in professional mobile radio (PMR) used for broadband data communication. The designed system shows better robustness to noise uncertainty and interference. A dual-stage SS algorithm for CR is designed in [124] using cosine modulated filter banks (CMFB), which have a simple design and real coefficients for the filter.

Compressed sensing is based on the presumption that the sparsity of the wideband spectrum is known, but due to the dynamic spectrum environment and dynamic nature of PU,

the sparsity is unknown. In [128], an adaptive compressed spectrum sensing scheme is proposed to find the optimal sampling rate and detect the spectral holes when the sparsity of the wideband signal is unknown or varying. The authors in [129] use a hybrid model by merging compressive spectrum sensing with geolocations to determine spectral holes in a decentralized CR. A dynamic compressive wideband SS is proposed in [130], which is based on recovering the channel energy of the spectral components, which show varying occupancy status in successive time slots. Bayesian compressive sensing with L_1 -norm minimization and with circulant matrix is proposed in [131] to improve the speed and efficiency of the recovery stage in compressive sensing.

Multi-coset sensing (MC) is a periodic non-uniform sub-Nyquist sampling method for obtaining sparse multiband signals. In [136], an adaptive blind wideband spectrum sensing for CR is proposed where the wideband SS is based on non-uniform multi-coset samplers, and the system adapts to variations in the number of active bands and input signals. The main advantage of multi-coset sensing is that each channel's sampling rate is much lower than the Nyquist rate. In [137] the problem of unknown band position is handled through discrete multi-coset sampling. A joint optimization

TABLE 5. Classification of cooperative sensing models.

Topology	Design model and References	Advantages	Disadvantages
Centralized	• A central node or fusion center chooses a frequency band, merges the collected lo- cal sensing data from all CR users, evaluates the existence of PU, and the decision is dis- seminated back to cooperating CR users [143], [144], [140].	 Efficient decision made by FC after complete local data acquisition. Can achieve multiple optimization objectives. Fewer message transmission. 	 Vulnerable to FC failure. More energy requirement for transmission if distance and number of nodes are high. If more nodes, the data collection phase is large. High-quality channel required due to CCC.
Cluster-based	• Division of the network into multiple clusters with each cluster comprising of a CH and several members.Nodes report their local decisions to CH which forwards the data to FC or makes a decision at cluster level, and reports it to FC [147], [150]–[152].	 Reduction in the time needed for delivering data by a factor of the number of CH. Mean data transmission dis- tance lower than centralized topology, so less transmit power required. 	• More consumption of energy at CH for processing of data.
Distributed	 No requirement of central node. Nodes exchange data in a determined order, then take global decisions based on data obtained from neighbouring nodes. 	 Data collected from surround- ing nodes. Reduced number of links. 	 Not efficient as only local data used to make decisions. At risk due to malicious activities and inaccurate data. Inefficient in terms of power consumption.
Relay-assisted	• Relaying the reported sens- ing messages by another node to the FC or a selected node where local sensing informa- tion depends upon the channel quality experienced by a node [68], [176], [177], [178].	 Message exchange time depends on the longest route of network but it is shorter than centralized topology. Distance among neighboring nodes is shorter than the average distance in a centralized case. 	 High energy consumption at relays due to data processing. Fast and efficient route selection mechanism required for less consume less energy and time.

algorithm is developed using an adaptive least coset number to recognize various features and choose an appropriate algorithm for signal reconstruction.

When the primary user's usage information is unknown to the CR, wideband spectrum sensing makes greater use of the frequency spectrum. However, the execution of WBSS is quite complicated as it has extensive usage of energy. The spectrum sharing stage aims to assign vacant bands to secondary users efficiently, without any collisions.

D. COOPERATIVE SPECTRUM SENSING

Single-node spectrum sensing provides us with unreliable performance due to deep fading and shadowing, which increases the rate of missed detection. CR receivers require high sensitivity, which in turn increases the implementation cost and complexity. Moreover, if primary users' SNR is below a certain threshold, increasing the sensitivity would not help in improving the detection performance [141]. To overcome these issues, cooperative sensing is used wherein the spatial diversity advantages are exploited to strengthen sensing performance. It is possible to distinguish cooperative sensing models into four categories: centralized, distributed, relay-assisted, and clustered [142]. Table 5 classifies various models in cooperative spectrum sensing with a brief description, their advantages, and disadvantages.

Centralized cooperative sensing consists of a fusion center (FC) which controls the stages of cooperative sensing. Omnidirectional antennas are used in most of the cooperative sensing techniques. In [143], to achieve fine-grained sensing and broader sensing range of PU, a directional antenna is used. The fusion center gathers the sensing information from SU and optimizes the sensing power, sensing period, and sensing beams for each SU with reduced energy consumption and maximized probability of detection. A reliability-based decision fusion scheme for cooperative spectrum sensing is proposed in [144], where the reliability of each SU decision is considered as a weight factor while deciding upon the ultimate decision at the FC. Probabilistic estimation is carried out by comparison with the past data about local and global decisions. In [145], integration of belief propagation (BP) algorithm with a fast detection technique is used

TABLE 6.	Comparison	between	different	spectrum	sensing	techniques.
----------	------------	---------	-----------	----------	---------	-------------

Harvesting Model	Design objective	System model	Advantages
Wavelet-based spectrum sensing [120]	Mitigate spectrum leakage and interference using WPSS, and reduction of sidelobes using FBMC	60 GHz mmWave cognitive 5G heterogeneous small cell networks with interference from WiGig system	Improved performance compared to conventional OFDM
Centralized cooperative directional sensing [143]	Optimized sensing parameters for each SU and maximize detection probability	FC forms multiple clusters and is equipped with directional antennas	Reduced energy consumption and better performance than non-optimized schemes
Multi-objective optimization using genetic algorithm for spectrum allocation [88]	Maximizing both network throughput and spectrum utilization	Multi-hop concurrent data flow CR for IoT	Constraints of mutual interference and resource competition in link concurrent transmissions is considered
Bayesian data clustering for dynamic SS [147]	Find spectrum opportunities in temporal, spectral, and spatial domains with faulty sensing data and correlation among SU measurements	N stationary SUs and $Mfrequency channels with CCCwith SUs and BS$	No requirement of prior knowledge of network topology, location of the sensors, or the number of clusters
FD-CRN with brute-force search and PSO algorithms [91]	Acheive optimal detection thresholds for SU spectrum sensing with throughput maximization	PU and the SU operate in a time-slotted manner with SU having self interference cancellation capacity	Higher throughput in conditions of low SNR or high transmission power for PU
Transferring reputation mechanism and dynamic game model based secure collaborative spectrum sensing strategy [149]	Decreases malicious attacks, resolves the reputation loss problem, and helps SUs to provide honest data	SU have CR with omnidirectional antenna in a CRN-CPS with energy sensing	Increases spectrum sensing accuracy and protects against the internal attacks
Multiband SS and cross-layer reconfiguration for dynamic resource allocation [119]	Optimization problem to find optimal number of channels to be assigned to each IoT node for SS	IoT applications in cognitive 5G networks	Fulfil QoS requirement by opportunistic spectrum access

for cooperative spectrum sensing in order to attain a better detection performance in terms of delay and false alarm rate, where PU actions are heterogeneous in space and dynamic in the time. After considering the practical and imperfect common control channel (CCC), the authors of [146] suggested a robust reputation based cooperative spectrum sensing technique where the impact of imperfect CCC on detection of malicious SU is studied and is verified to be resilient against attacks. In [147], a Bayesian data mining framework is proposed for clustering SU based on their observations in a spectrum heterogeneous CR network. Each SU's reliability and spectrum occupancy are inferred, and a multi-label graph cuts method is proposed to find out spectrum opportunities. Cooperative spectrum sensing performance gets deteriorated due to the existence of malicious interference in CRNs. In [148], to tackle malicious interference, a new cooperative spectrum sensing detector is proposed to estimate the local signalto-interference-plus-noise-power ratios (SINRs) at secondary users and then transmit it to a fusion center for making the decision. To counter the potential security risks from internal nodes, in [149], a location-aware transferring reputation mechanism and dynamic game model-based secure collaborative spectrum sensing strategy is presented. This helps to solve the reputation loss by user mobility, motivate SUs to provide honest information, and increases the precision of spectrum sensing in CR network-based cyber-physical

VOLUME 9. 2021

systems (CRN-CPS). Table 6 compares various spectrum sensing protocols discussed in literature.

E. MERITS AND DEMERITS OF SH AND CR

CR in opportunistic mode senses the unused or underutilized EM spectrum to harvesting bandwidth. These licensed frequency bands are allocated to primary users, which include TV, satellite, and beyond. While the SH technologies are critical, bandwidth resources are limited, whereas communication networks are many. With reliable and efficient SH technologies, it is possible to mitigate the spectrum scarcity problem. While SH technologies do possess several merits, they also have demerits, for example, implementation issues. Below, we present a brief summary of these merits and demerits that may vary from one SH approach to another.

Demerits: SH technologies have certain demerits, mainly in terms of implementation issues. Some of these are listed below. In opportunistic CRs, accurate spectrum sensing is the most challenging task. The false negatives during detection would result in simultaneous communication by primary and secondary users, thus interfering with licensed users.

 i) SH-enabled CR improves the complexity and cost of intelligent systems. This cost and implementation issues vary drastically from one cognitive network to another, such as terrestrial and satellite cognitive networks and ad-hoc cognitive networks. ii) The complexity of energy harvesting and bandwidth conserving CR is even more. For example, the harvested energy has to be utilized for spectrum sensing as well as data communication.

IV. HYBRID SYSTEMS: PERSPECTIVES AND REMARKS

This section provides a brief overview of some of the hybrid systems that employ the energy and/or spectrum harvesting techniques.

Next generation EH wireless systems and EH-IoT: We can broadly classify next-generation EH communication [38], [159] systems into three categories: non-EH wireless systems, EH, or fully autonomous wireless systems, and hybrid or partially autonomous wireless systems, including IoT and cyber-physical systems. The first category of wireless systems, non-EH systems, needs dedicated and reliable power sources. In the second category, EH wireless systems, there is no need for a dedicated power source such as a battery. These wireless systems are fully autonomous because they can harvest energy from ambient sources and perform sensing and communication tasks. However, intermittent communication is a potential risk if the ambient energy sources are not available or sufficient to harvest the required energy. Therefore, these wireless systems are not useful for some critical applications such as broadcasting.

It is more practical to design and develop hybrid wireless systems and EH-IoT that consists of some fraction of EH nodes. The protocol stack should work in such a way that each layer's protocol(s) is(are) aware of the node or terminal's energy consumption and delivers optimal performance by utilizing power adaptively. Modifying the existing protocols such as artificial intelligence (AI)-enabled protocol stack can provide superior performance in complex, hybrid, next-generation wireless systems. The AI-enabled stack should also include a dedicated environment layer (or its equivalent) in addition to layers such as infrastructure, perception, computation, storage, and application. The environment layer may consist of protocols that could process and communicate data related to the carbon footprint by the AI-enabled hybrid cyber-physical system (CPS) or hybrid cyber-biophysical system (CBPS) or network. It could also be responsible for communicating useful data (environmental conditions, machine health, etc.) to the remote server during emergencies (like natural disasters). Furthermore, this layer could improve the AI-enabled CPS or CBPS' protection and security in which the AI stack is mounted. Since AI is being applied (and would be applied) in several complex systems in various fields of science, engineering, business, and health, the AI stack needs to be greener and eco-friendly.

Next-generation SS/SH wireless systems and SS/SH-IoT: We can broadly classify next-generation communication systems into three categories: non-spectrum sharing wireless systems, SS/SH wireless systems, and hybrid or partially spectrum sharing wireless systems, including IoT or CPS. The first category of wireless systems, non-spectrum sharing systems, needs dedicated bandwidth resources always—for example, security and surveillance satellite links. In the second category, SS/SH wireless systems, there is no need for dedicated bandwidth. The nodes are intelligent and harvest bandwidth opportunistically from the unused EM spectrum of primary users. These secondary SS/SH wireless systems also have limited applications in which continuous transmission of information is not needed.

It seems more practical to design and develop hybrid SS/SH wireless systems and cognitive IoT that consists of some fraction of primary nodes with dedicated bandwidth allocation and others being secondary. The protocol stack of such hybrid cognitive networks should work in such a way that each layer's protocol(s) is(are) aware of the node or terminal's bandwidth consumption and delivers optimal performance by utilizing limited harvested bandwidth in the best possible manner.

Next-generation EH-SS/SH wireless systems and EH-SS/SH-IoT: It is quite challenging to develop a protocol stack for such complex wireless systems. With the advent of AI tools and algorithms, it is possible to build a protocol stack in which each layer is aware of energy consumption and bandwidth utilization. As mentioned before, it is a good idea to first design hybrid next-generation EH-SS/SH wireless systems and IoT before making them autonomous. Depending entirely on EH nodes whose performance depends on ambient energy sources, makes it hard to achieve better reliability, quality of service (QoS), and quality of experience (QoE). However, using multi-energy harvesting systems and efficient combining of these EH modules could enable the complex EH-SS/SH-IoT to utilize more green and clean energy than non-renewable energy. Thus, the hybrid and optimal designs would provide a balance while reducing the carbon footprint significantly over time.

Figure 12 illustrates a CR network with energy harvesting. A fading channel with additive Gaussian noise is assumed and data and energy queues are present with the transmitters in which data packets and harvested energy or arriving energy are stored, respectively. SUs do both information transmission in the unused spectrum and RF energy harvesting in the spectrum occupied by PUs. Integration of energy harvesting technologies and bandwidth conserving CRs could provide even more benefits to the next generation of communication systems, networks, cyber-physical systems, etc. For instance, in EH-CR-IoT, the IoT-sensors could leverage ambient renewable resources (e.g., RF energy resource) for energy harvesting and the harvested bandwidth for communication. However, there are various implementation challenges and open problems such as design and development of efficient spectrum, and energy-aware, and optimal protocols for dynamic EM spectrum access and harvested spectrum management. Furthermore, the cost and complexity of the EH-SH CR system are other significant challenges that need to be addressed by researchers and industries.

Numerous technologies such as cooperative communication and CRs are suggested to resolve the constraints of two valuable assets, energy and spectrum, and lack of



FIGURE 12. The model of CR network with energy harvesting [20].

performance due to wireless fading channels. Communication with cooperation has generated a great deal of interest because of its ability to minimize the fading due to wireless channel and enhance wireless links' performance by leveraging the spatial diversity benefits inherent in multi-user scenarios [120]. As nodes communicate with one another to transmit the information, they form an invisible multi-input multi-output (MIMO) system. Nevertheless, as we know that wireless nodes are mostly constrained in terms of size, these nodes require batteries with longevity and large storage capacity, so EH methods have been introduced to enhance the throughput without the need for replacement of batteries periodically.

Medium access control in cooperative and cognitive wireless systems plays a crucial role in sensing the channel, mobility of spectrum, allocating the resources, and sharing of spectrum. MAC design is affected by sensing duration and precision with an extra burden of optimizing the energy consumption for better system performance and QoS. In [156], a cooperative MAC protocol or cooperative cognitive TDMA is proposed for cognitive networks that does opportunistic EH in SUs. It uses both standard and spectrum leasing methods to achieve QoS for PUs and increases throughput.

A. ENERGY EFFICIENT RESOURCE MANAGEMENT

Energy consumption in wireless CRN is also affected by the radio resource management techniques. A summary of the energy management and related IoT problems is provided in [160], two types of IoT management are classified. Energy-efficient remedies include optimum sleep and idle state scheduling. Repeated scanning of the spectrum for efficient protection of PU results in wastage of energy. In [161], EH-SS based cognitive radio sensor networks (HCRNs) are proposed to deal with the problem of allocating the resources in HCSRNs. Scheduling is done for spectrum sensors to maximize the mean accessible time of channels and resource allocation, i.e., power, channel, and time for data sensors for minimizing its energy consumption. For example, in [162] an energy-efficient spectrum sensing heterogeneous cognitive radio sensor network (EH-CSRN) is considered. The proposed technique focuses on reducing the implementation cost, ensuring the sensing performance, and detection of usable channel time for the transmitting data by using a SS algorithm. It is based on cross-entropy to identify channel assignment scheme which reduces the deployment cost of network equipment, thereby promoting green communication.

In [164], physical-layer security for EH-aided CR is proposed. Two scheduling algorithms that are advantageous for the users are described as channel-aware user scheduling (CaUS) and energy-aware user scheduling (EaUS). In CaUS, a user with maximal channel gain is selected for cooperative transmission users. In [165], time scheduling for a backscatter-aided RF-powered CR network is proposed, where multiple SUs transmit information to the same secondary gateway, which then allocates the time-resources in the backscatter mode and the harvest-then-transmit mode. For conducting time resource management, two auction-based time scheduling methods have been suggested. In [163], a heterogeneous cognitive radio network (CRN) that minimizes energy consumption due to sensing, reporting, and channel switching per unit by a joint objective function for spectrum sensing and energy efficiency is proposed. An energyefficient resource allocation method is proposed in [166] for energy harvesting-based cognitive machine-to-machine (EH-CM2M) communication to control the limitations due to spectrum and battery capacity. The joint optimization of channel selection, peer discovery, power control, and allocation of time is done to maximize M2M transmitters' energy efficiency. Resource allocation in wireless powered wideband CR under a practical non-linear EH model is considered in [167] to jointly optimize EH time, transmit power, channel allocation, and maximizing sum throughput of SUs.

An optimal resource allocation is proposed in [168] to maximize the sum energy harvested and max-min fairness harvested energy in a wideband sensing-based CR network with SWIPT, subject to limitations on transmit power, rate, sub-channel assignment, and interference. A resource allocation in multi-carrier CR is proposed in [169], where PUs are powered by SUs in a cooperative manner. In this scheme, SU can concurrently transmit both its own data as well as transfer data to PU where harvest-then-transmit for synchronizing WPT and WIT. First wireless power transfer is done to PU, then two methods, subcarrier sharing and subcarrier exclusivity, are used for data transfer and resource allocation algorithms are used to maximize the rate of SU under limitations to achieve minimum PU rate and SU transmit power.

B. ENERGY EFFICIENCY VIA CR

In wireless communications, energy availability is often a significant problem, and SUs have to expend extra energy to identify several channels and continuouly switch between various channels. Thus, energy efficiency and spectrum efficiency are key aspects in CR. Currently, the battery is the primary energy source. However, limitations of battery life, periodic replacement, and environmental pollution have led researchers to look into sustainable, environmentally friendly power supply.

In [170], the authors propose a CR system that harvests energy from the RF signal of PU during sensing time and detection time. An optimal sensing time is estimated, which maximizes the harvested energy and average throughput of the system. In [171], a wideband CR with SWIPT based EH is proposed, where max-min fairness in the allocation of resources between users is studied. Further, opportunistic accessing of the spectrum with imperfect CSI with maximization of throughput of the worst-case SU by performing a joint optimization of the sensing time is studied. In [172], a utilityoptimal resource management and assignment scheme for cognitive radio based sensor networks (CRSNs) with EH is proposed with continuous supply energy for facilitating the power-constrained sensors. An online framework for managing the energy and spectrum resources and distribution of resources is proposed by Lyapunov optimization. Three stochastic processes are handled by this approach: dynamics of energy harvesting, channel utilization inaccuracy, and channel fading. The algorithm fulfills two significant objectives: first, regulating the harvest and use of energy by sensors while maintaining the queues for energy and data, and second, optimization of the use of spectrum which is licensed while preserving a bearable collision rate among the unlicensed and licensed users. A slotted mode CR system is presented in [173], in which SU is powered by EH from an ambient environment. The mechanism considers EH and SS in three non-overlapping time-slots using data fusion and decision fusion and focus on the "harvesting-sensing-throughput" by considering multi-slot spectrum sensing paradigm and maximizing SU's throughput and finding out optimal sensing duration and threshold. Based on the sensing effects,

mode while transmitting to maximize throughput, and continue in sleep mode to preserve energy, or extract energy from PUs. A partially observable Markov decision process method is suggested to evaluate the mode of transmission with a threshold on energy, and choose between SS or sleep mode depending upon actions of PU. With the increase of throughput, there is also an increase in SU's computational complexity, latency, and the system's energy demand. A twoway information exchange dynamic SS framework is proposed in [175] to enhance energy efficiency for transmitting data in licensed channels. Also, an energy-efficient optimal transmit power allocation scheme is derived to improve dynamic SS and data throughput. A novel non-orthogonal multiple access (NOMA) assisted overlay spectrum sharing for multi-user CRNs for improved spectrum utilization is proposed in [176]. Scheduling of a SU, which acts as a relay, is done to transmit both its own and PU signals using NOMA based cooperative spectrum sharing. Two scheduling schemes: reliability oriented secondary user scheduling (R-SUS), which helps in minimizing PU and SU outage probabilities, and fairness oriented secondary user scheduling (F-SUS) achieves superior user fairness are used, which achieves full diversity order.

in [174], the SU can work in both overlay or underlay

Outage performance analysis is studied in [177], [178] for wireless EH underlay Relay-Assisted CR networks with node position taken into consideration. Based on stochastic geometry, outage probability with and without relay-assisted transmission is derived, and selection of the optimum relay based on the comparisons of above-derived probabilities is made. In dominant interference and noise environments, a spectrum sharing scenario is considered in [179], where the SUs communicate via an interference-aided EH DF relaying with multiple primary transmissions and outage limitations. Relays use time-switching to shift between EH and data transfer. In the EH scenario, the advantage due to harvesting energy from primary interference, which acts as an energy source, suppresses the negative effects of primary interference and helps to maximize the secondary throughput. Also, using multiple antennas at the secondary relay aides in EH. In [180], an overlay spectrum sharing is considered for SWIPT based EH. By employing decode-and-forward relays for assistance during PU transmission under Nakagami-m fading channel, outage probability expressions for PU and IoT devices are derived. EH is done by IoT devices with RF signals from PU with maximizing system throughput and energy efficiency. In [181], joint optimization of EH and SS is done with limitations on energy causality, collision, and temporal correlation of channel sensing probability and mean channel capacity as the realizable throughput. Maximized throughput and optimum detection threshold is obtained, where the primary channel state is a discrete Markov process. In [182], a green CR that operates in a harvesting-transmitting fashion within each time slot is proposed where CR nodes harvest energy first then transmit data using the harvested energy.

C. EH USING COOPERATIVE COMMUNICATION

CR implementation with cooperative communication makes it possible for the SUs to get more chances for transmission by acting as a relay for PUs. In [183], the authors consider SU with ambient EH and infinite cell capacity that can possibly cooperate with PU for EH duration and relaying power to maximize throughput. The authors in [184] considered a CR system where SU creates a limited capacity energy and unlimited capacity data queues to store its own and PU's data packets which are not delivered successfully. SU uses the Alamouti coding scheme for cooperation with PU, which has queuing delay limitations; after that, throughput maximization of SU is carried out subject to constraint of all data queues' stability.

In [185], a multichannel CR is suggested to improve the performance. At the same time, the transmitting SU divides the subchannel into two slots, the first one for sensing the PU by cooperative SS and the other for harvesting PU's RF energy by multichannel EH. By jointly optimizing the subchannel set, the time required for sensing and power used for transmission, throughput, harvested energy, and effective energy utilization by SU is maximized. A collaborative scheme for sensing the spectrum (CSS) in EH-CWCPN is proposed in [186]. The clustering of CR nodes is done based on the received signal's power levels for increasing the utilization of the spectrum. The time slot is shared between the harvesting of energy, sensing the spectrum, and information transmission. The optimization problem aims to maximize the throughput with energy and collision limitations under a detection threshold, which is obtained by formulating a fictitious cognitive node and number of clusters of cognitive nodes. Cooperative sharing model for CWPCNs with both amplify-and-forward (AF) and decode-and-forward (DF) protocols to maximize its uplink energy efficiency by using an optimization problem and then dividing it into two parts, uplink scheduling and cooperative power control, has been proposed in [187]. In [153], the cognitive sensors sense licensed channels and find out PU's availability by computation over multiple-access channel (CoMAC) soft-decision cooperative spectrum sensing method. Energy-efficiency maximization is done for CSNs by using the time required for sensing, the detection threshold, and the length of symbol sequence as variables. In [188], a SWIPT-enabled cooperative network is proposed where a relay helps in PU's transmission. Here, SUs acting as relays superimpose their data on PU's information. SWIPT relay uses an energy-assisted decode-and-forward (EDF) protocol. A relay-based spectrum sharing protocol in CR is proposed in [189]. SU does RF-energy harvesting from PU in exchange for helping out in data transmission. Alamouti coding is used for primary data relaying. Superposition coding techniques are adopted for secondary data transmission in a continuous manner with joint decoding and interference cancellation methods employed at receiver. Finally, optimal power assignment is done to maximize SU throughput and guaranteed PU QoS.

A dynamic way of accessing the channel has been suggested in [190] for clustered CR nodes where maximizing the resource distribution by finding out conditions for both inter and intra clusters is proposed, and sensing and switching to a licensed spectrum by a sensor is performed to increase energy efficiency and maximum utilization. Due to limited spectrum and increasing wireless devices and cognitive radio, a new cognitive wireless powered communication network (WPCN) is proposed in [192]. In the proposed WPCN, devices are powered via wireless energy transfer (WET) by a hybrid access point in the downlink and does wireless information transmission (WIT) in the uplink. Sum-throughput maximization is done for cognitive WPCN by jointly optimizing the time assignment for the harvest-then-transmit protocol and the WET/WIT transmit power. Both underlay and overlay-CR are used to maximize the sum throughput for wireless energy and information transfer by optimizing different transmission limitations. Figure 13 shows a WPCN network where W amount of time in each block is for WET, and the remaining time (T-W) is for WIT. In [191], a doubledeck cluster cooperative relay assistance model in hybrid spectrum sharing CR networks is proposed where energy efficiency is achieved by optimizing SUs for the upcoming 5G networks.

D. APPLICATIONS IN DIVERSE FIELDS

With the advent of 5G, IoT, and its variants such as IoMT, IoBNT so on, there are promising potential applications of EH and SH in various cyber-physical systems and nanoscale networks. Some of these applications are listed below. First, we mention EH applications followed by SH and EH-SH applications.

SWIPT-enabled wireless networks: EH radios can harvest energy from ambient renewable sources such as RF signals and also simultaneously transmit information. The harvested energy can be used instantaneously or stored in a supercapacitor for future communication, depending on the application. For example, low power sensors-equipped body area network (BAN), and wearable devices for smart health could exploit techniques like SWIPT. With proper energy management, these networks could be more energy efficient or greener.

Ambient backscatter communication systems: AmBSC comes under passive IoT. The ambient backscatter devices (AmBSD) such as EH tags can harvest energy from ambient RF signals such as TV, Wi-Fi, etc. The number of EH rectennas varies from one type of AmBSC to another. Intuitively, it is expected that more number of EH rectennas can scavenge more ambient energy. However, as the number of rectennas increases, the size and complexity of the AmBSC would also increase.

Cooperative D2D communications: EH relays are useful and can deliver superior performance when compared to D2D communication without relays if designed optimally. The performance benefits include increased data rates, network throughput, network lifetime. selection of the type of relay. Its optimal design depends on various aspects such as power



FIGURE 13. A cognitive WPCN with cooperation between primary and secondary systems with the harvest-then-transmit protocol for WPCN [192].

and bandwidth constraints, acceptable complexity, to name a few.

Wireless sensor networks (WSNs)/IoT: IoT focuses on making the Internet universal and omnipresent and can impact many aspects that affect the standard of living of consumers. Technologies intended for IoT cover a broad range of areas comprising cyber-physical systems, automation at home, smart healthcare, smart cities, monitoring, intelligent ecosystems, smart transportation, and many more. The implementation of IoT devices, therefore, will provide substantial savings and revenues through the provision of control and management services. This acceleration has forced organizations to implement various methods to address this rapid growth and development and the problems connected with them. These include: allocation of sufficient spectrum bands in IoT, improving transmission efficiency, scarcity of spectrum, sharing of spectrum, interference among devices, etc. For example, a wireless sensor network (WSN) usually comprises of many affordable sensor nodes. Such nodes have restricted battery power capacity. When the node's battery drains, it should be restored as the network may not function correctly if adequate nodes are not properly working. A feasible approach for extending the lifespan of wireless energy-limited networks is the energy harvesting methodology that has lately gained tremendous attention. It allows wireless nodes to scavenge energy from the local environment. In scenarios where the frequent replacement of batteries is infeasible, EH technologies are a suitable alternative. However, if deploying EH sensors is not feasible, hybrid sensors could be useful, which harvest energy whenever available and as a backup, depend on battery energy. These hybrid sensors could enhance battery life as well as the WSN life period.

In wireless sensor networks, nodes typically transmit whenever an event of interest occurs. Since the continuous transmission is not required, the cognitive radio-enabled nodes can occasionally harvest spectrum and transmit data. This way, no dedicated bandwidth is necessary for these networks. Similarly, CR-enabled IoT communications over the harvested spectrum will improve spectrum utilization. However, various technical challenges and open problems such as spectrum sensing, spectrum allocation, dynamic spectrum access and management, modeling, and simulation of complex cognitive IoT environments need to be addressed.

5G and beyond: Next-generation 5G/6G mobile networks could use EH technologies in some of their network components. For instance, the hybrid base station of eNodeB/gNodeB partially depends on harvested energy. The inclusion of hybrid EH network entities could make the 5G/6G networks greener and reduce their carbon footprint.

Applications of spectrum harvesting/sharing systems: Though there exist some unlicensed frequency bands (ISM bands), a majority of the frequency bands are licensed. As the spectrum is precious, it cannot be underutilized. As mentioned before, spectrum sharing/harvesting systems are useful in addressing these issues. Below, we present some of the applications of spectrum sharing or cognitive radio systems.

a. *Collaborative or cooperative spectrum sharing systems:* The application of cooperative relaying techniques enhances secondary user performance for devices that opportunistically sense the primary user's spectrum and harvest it. However, a secondary underlay user can only transmit while satisfying the constraints imposed on transmissions. These are interference constraints which the limit secondary's performance. Cooperative relays adaptively adjust the parameters (e.g., gain) to optimize the end-to-end secondary performance in terms of error probability, spectral efficiency, and energy efficiency.

- b. *Passive IoT of AmBSC:* In addition to backscatter tags, in non-reliable links, adaptive relays can be exploited to improve the AmBSC system performance. Incorporating both battery-less tags and efficient relays could enhance the reliability and data rates of AmBSC. The optimal placement of these tags and relays is an open problem and can be explored by researchers.
- c. Emergency communications: Professional mobile cognitive radios could sense and exploit the unused EM spectrum for emergency communications and public safety communications. These radios are useful for emergency services and could be used by ambulance, fire-station staff, and police, to name a few. Limited efforts in industry and research have been conducted to use small cell base stations installed on drones for emergency communications [199]. It is possible to use such flying base stations in disaster zones, emergencies, or in rural areas. Data security [200] is the primary obstacle in implementing such flying base stations. It is difficult to deploy complex security mechanisms with drones as there are resource-constrained devices. Furthermore, in implementing such networks, the aspects of user affiliation, drone operation, and bandwidth allocation are major bottlenecks. Thus, in such cases, energy harvesting and spectrum harvesting can be a very useful tool.

The applications mentioned above are also valid for EH-SH systems. However, the performance of these systems is highly limited due to harvested energy and bandwidth limitations. These systems are more suitable for low power IoT applications in which nodes have minimal harvested energy and transmit at low data rates. Energy-aware optimal PHY and routing protocols need to be implemented to optimize their performance.

V. FUTURE RESEARCH TRENDS

In this comprehensive survey, we reviewed EH and SH from different perspectives. We now summarize the key themes and the lessons. The rapid growth of feasible and implementable ESH techniques would lead to their application in various systems and networks from an economic perspective. Further, the standardization of these technologies in next-generation IoT and cyber-physical systems would enable rapid manufacturing and production. Thus, these technologies will eventually appear in the market as commercial products for various life needs, such as infotainment, smart health, smart transport services, etc. However, several obstacles exist in making this a reality due to several critical issues like security, privacy, and other technical challenges like poor or low efficiencies. For instance, most green energy harvesting approaches can only deliver limited power, limiting their applications. Similarly, spectrum harvesting methods'

success is also limited due to the highly random and dynamic nature of the primary users, causing dynamic spectrum access and management exceptionally challenging. Incorporating automation with artificial intelligence, machine learning, or deep learning wherever feasible and applicable would address this problem in more complex IoT or cyber-physical systems. Furthermore, the performance of EH and SH techniques and protocols implemented in secondary users or terminals will drastically vary depending on the mode of the operation and type of communication system or network. In addition to this, there are compatibility, interoperability, and scalability issues that need to be addressed in more complex systems that exploit EH and SH techniques.

Some of the EH related challenges and issues may find their solutions in the form of smartly controlled energy harvesting, integrated storage, and efficient transmission approaches. Similarly, some of the SH related challenges could be resolved using hybrid and flexible modes of operation of spectrum sensing and sharing systems and networks. Various proprietary technologies (e.g., ZigBee and Bluetooth) should come forward and work together to alleviate issues such as interoperability, compatibility, and scalability. This paper presented various EH and SH-enabled network architectures, techniques, protocols, and EH and SH standards. The optimization and practical implementation of these green systems and networks under various power and energy constraints form the lion's share of further research in EH, SH, and ESH communication technologies.

This paper also discussed various recent investigations in order to highlight technical challenges and open problems in areas such as green cognitive IoT and its variants such as energy-efficient xG, and M2M communications. Especially for EH and SH enabled IoT, there are issues such as privacy and security. Novel algorithms with moderate complexity to improve end-to-end performance in energy efficiency, spectral efficiency, reliability, and security for various IoT applications are very much needed. Smart and hybrid computing and communications techniques and algorithms would improve the ESH-enabled IoT and other systems and network performance. However, these techniques or algorithms should be feasible and practically implementable.

Future technologies need to be green as much as possible. Eco-friendly communication technologies and IoT systems and its variants should replace power-hungry low energyefficient, and low spectral-efficient systems and networks. Overall, despite the growth of EH and SH technologies in research for green communication and networking, their visibility in various commercial wireless networks and IoT is not significant. There exists a significant gap between analytical EH and SH techniques and protocols, and their implementation in practice. This gap needs to be minimized as much as possible. This bridging of theory and practice seems possible with out-of-the-box thinking and developing next-generation miniaturized electronics, as well as robust and secure algorithms based on intra-disciplinary and interdisciplinary approaches wherever possible.

A. COMMERCIAL AND ECONOMIC PERSPECTIVES

With the advent of feasible EH and SH techniques and technologies, manufacturing industries could find ample growth opportunities to develop new heterogeneous sensors and cognitive devices. These EH and SH enabled sensors and CRs would bring more revenue for various sectors related to IoT, 5G, defense, aviation, consumer electronics, etc., Figure 14 depicts the predicted EH electronic devices market scenario [33]. The growth of the EH and SH components or devices market will lead to the economic development of various autonomous cyber-physical systems and next-generation wireless systems and networks. Figure 15 shows the projected SH or cognitive radios market. From the bar chart, we can see that there is approximately an increase of 164% in the CR market. The CR market may be viewed in two segments, namely, component and application. While the former segment includes antennas, RF, or millimeter wave (mmWave) electronics, the latter segment includes cognitive radio networks, CR-IoT, ad hoc networks, to name a few. Based on [24], the CR market is expected to be worth 7.44 billion USD by 2022. In it, the CR market mainly comprises of components such as software tools and hardware; the application consists of spectrum sensing, spectrum access, cognitive routing, etc. The statistics depicted pictorially in Figure 14 and Figure 15 reveal that significant and rapid growth in SH and EH devices and systems can be expected in various CPS and communication networks.



FIGURE 14. EH market scenario (2011-2021).

B. TECHNOLOGY READINESS LEVEL

For the energy and spectrum efficiency research, in this section, we present a technology readiness level (TRL) diagram, as shown in Fig. 16, which maps the output levels of the research across different levels of maturity. The TRL is divided into three stages with TRLs level 1-2 denoting the fundamental research in the area, level 3-5 denoting the technology development, and level 5-6 denoting the technology demonstration in practical networks. The research in EH and SH technology has matured significantly with the greater push and awareness on the green communication networks and further, due to the focus on energy optimization. Many of the present communication networks and the future



FIGURE 15. CR market scenario (2017 to 2024) [19].

wireless networks are envisaged to be green networks with energy efficiency as one of the desired characteristics. On the other hand, due to spectrum scarcity and an increase in the number of users and diverse fields of applications, spectrum efficiency is a desirable feature. The TRL diagram gives a pictorial representation of the development of the EH and SH technology from the fundamental research level to the present and future aspects of implementation.

C. TRADEOFFS FOR GREEN NETWORKS

Future wireless communication networks, specifically 5G and beyond 5G networks, have a greater push towards efficient green communication, which is seen in both academic and industrial research. For an AWGN channel, the tradeoff between the spectrum and energy efficiency can be expressed as [157],

$$T_{ES} = \frac{T_S}{(2^{T_S} - 1)N_o},$$
 (11)

with

$$T_S = \log_2\left(1 + \frac{P}{BN_o}\right),\tag{12}$$

where P denotes the total transmit power, W denotes the total bandwidth, and N_o denotes the noise power density.

The present communication networks have orthogonalfrequency-division multiplexing (OFDM) as the key enabling technology, which forms the basis for 4G systems: For future communication networks, which include 5G and beyond 5G, new technologies which are more efficient than OFDM are being explored. The non-orthogonal access technologies studied in the literature that are considered as prime candidates for future network deployments are D2D and non-orthogonal multiple access (NOMA) [23], with both the techniques focusing on inter-user interference. The capacity of non-orthogonal access technologies, as described above, can be expressed considering all the transmitter-receiver pairs as [154],

$$C = B \mathbb{E}_{h_{p}, h_{p}^{m}} \times \left[\sum_{p=1}^{N_{t}} \log_{2} \left(1 + \frac{P_{p} ||h_{p}||^{2}}{B N_{o} + \beta \sum_{m=1, m \neq p}^{N_{t}} P_{m} ||h_{p}^{m}||^{2}} \right) \right]$$
(13)



FIGURE 16. TRL diagram for EH and SH technologies for the next generation wireless communication networks.

where N_t is the total number of transmitter-receiver pairs, h denotes the channel gain, and β denotes the equivalent capability of interference cancellation. In comparison to the orthogonal schemes, the non-orthogonal schemes are able to achieve better spectral efficiency values. Further, the non-orthogonal schemes do not require additional costs as they do not need any significant hardware improvements over the existing infrastructure [49].

The implementation of energy-efficient techniques for future wireless communication networks needs to address several challenges, like whether the schemes be applied in a centralized or decentralized fashion. The answer to this lies in the way the infrastructure is utilized, and how the protocols are implemented. Future networks need to follow an approach in terms of maximum infrastructure utilization so that there is less strain on the resources For instance, the processing at the baseband level may be centralized, and the radio resource functionality may be divided across the whole network.

VI. CONCLUSION

The recent past has witnessed the rapid growth of heterogeneous sensors in IoT and proposals for architectures for their integration with the fifth-generation (5G) networks and beyond. Energy harvesting and spectrum harvesting technologies are essential to meet the requirements of two precious resources: energy and bandwidth for reliable and efficient communication. This survey paper provided a comprehensive overview of the EH and SH techniques for the next generation of wireless systems and networks. We presented an overview of the present trends of the EH and SH technologies, the implementation challenges, and open research areas. Specifically, we covered the techniques like the ambient RF-EH techniques and presented the spectrum harvesting techniques like those based on CR and cooperative sensing. Further, we introduced the TRL of the harvesting technologies and gave an overview of the future trends and challenges in the context of next-generation wireless networks. The EH and SH technologies have become an integral part

of the next-generation wireless communication networks' design and planning. This comprehensive survey provided various themes, perspectives, methods, and protocols that would serve as a useful resource for researchers in the field of advanced green and energy-efficient cognitive wireless systems and networks.

REFERENCES

- N. Zhao, S. Zhang, F. R. Yu, Y. Chen, A. Nallanathan, and V. C. M. Leung, "Exploiting interference for energy harvesting: A survey, research issues, and challenges," *IEEE Access*, vol. 5, pp. 10403–10421, 2017.
- [2] M.-L. Ku, W. Li, Y. Chen, and K. J. R. Liu, "Advances in energy harvesting communications: Past, present, and future challenges," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 2, pp. 1384–1412, 2nd Quart., 2016.
- [3] V. Chamola, B. Krishnamachari, and B. Sikdar, "An energy and delay aware downlink power control strategy for solar powered base stations," *IEEE Commun. Lett.*, vol. 20, no. 5, pp. 954–957, May 2016.
- [4] H. Ding, Y. Fang, X. Huang, M. Pan, P. Li, and S. Glisic, "Cognitive capacity harvesting networks: Architectural evolution toward future cognitive radio networks," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 3, pp. 1902–1923, 3rd Quart., 2017.
- [5] G. I. Tsiropoulos, O. A. Dobre, M. H. Ahmed, and K. E. Baddour, "Radio resource allocation techniques for efficient spectrum access in cognitive radio networks," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 1, pp. 824–847, 1st Quart., 2016.
- [6] Y. Liu, Y. Zhang, R. Yu, and S. Xie, "Integrated energy and spectrum harvesting for 5G wireless communications," *IEEE Netw.*, vol. 29, no. 3, pp. 75–81, May 2015.
- [7] V. Hassija, V. Gupta, V. Chamola, and S. Kanhare, "A blockchain-based framework for energy trading between solar powered base stations and grid," in *Proc. 21st Int. Symp. Theory, Algorithmic Found., Protocol Design Mobile Netw. Mobile Comput.*, Oct. 2020, pp. 315–320.
- [8] I. U. Din, S. Hassan, A. Almogren, F. Ayub, and M. Guizani, "PUC: Packet update caching for energy efficient IoT-based information-centric networking," *Future Gener. Comput. Syst.*, vol. 111, pp. 634–643, Oct. 2020.
- [9] K. Haseeb, I. U. Din, A. Almogren, and N. Islam, "An energy efficient and secure IoT-based WSN framework: An application to smart agriculture," *Sensors*, vol. 20, no. 7, p. 2081, Apr. 2020.
- [10] Cisco. (2020). Cisco Annual Internet Report—Cisco Annual Internet Report (2018–2023) White Paper. Accessed: Nov. 4, 2020. [Online]. Available: https://www.cisco.com/c/en/us/solutions/collateral/executiveperspectives/annual-internet-report/white-paper-c11-741490.html
- [11] M. H. Shahid, A. R. Hameed, S. U. Islam, H. A. Khattak, I. U. Din, and J. J. P. C. Rodrigues, "Energy and delay efficient fog computing using caching mechanism," *Comput. Commun.*, vol. 154, pp. 534–541, Mar. 2020.

- [12] A. Miglani, N. Kumar, V. Chamola, and S. Zeadally, "Blockchain for Internet of energy management: Review, solutions, and challenges," *Comput. Commun.*, vol. 151, pp. 395–418, Feb. 2020.
- [13] A. Jindal, N. Kumar, and M. Singh, "Internet of energy-based demand response management scheme for smart homes and PHEVs using SVM," *Future Gener. Comput. Syst.*, vol. 108, pp. 1058–1068, Jul. 2020.
- [14] M. Wazid, A. K. Das, N. Kumar, and J. J. P. C. Rodrigues, "Secure threefactor user authentication scheme for renewable-energy-based smart grid environment," *IEEE Trans. Ind. Informat.*, vol. 13, no. 6, pp. 3144–3153, Dec. 2017.
- [15] A. Jindal, A. K. Marnerides, A. Gouglidis, A. Mauthe, and D. Hutchison, "Communication standards for distributed renewable energy sources integration in future electricity distribution networks," in *Proc. IEEE Int. Conf. Acoust., Speech Signal Process. (ICASSP)*, May 2019, pp. 8390–8393.
- [16] R. Rajesh, V. Sharma, and P. Viswanath, "Information capacity of energy harvesting sensor nodes," in *Proc. IEEE Int. Symp. Inf. Theory*, Jul. 2011, pp. 2363–2367.
- [17] K. Kumar and Y.-H. Lu, "Cloud computing for mobile users: Can offloading computation save energy?" *Computer*, vol. 43, no. 4, pp. 51–56, Apr. 2010.
- [18] Mordorintelligence. (2020). Energy Harvesting Systems Market | 2020-2027 | Industry Report | COVID Insights. Accessed: Nov. 4, 2020. [Online]. Available: https://www.mordorintelligence.com/industryreports/energy-harvesting-system-market
- [19] Zionmarketresearch. (2020). Free Analysis: Intelligent and Cognitive Radio Market. Accessed: Nov. 4, 2020. [Online]. Available: https://www.zionmarketresearch.com/market-analysis/intelligentand-cognitive-radio-market
- [20] F. Hu, B. Chen, and K. Zhu, "Full spectrum sharing in cognitive radio networks toward 5G: A survey," *IEEE Access*, vol. 6, pp. 15754–15776, 2018.
- [21] A. Ahmad, S. Ahmad, M. H. Rehmani, and N. U. Hassan, "A survey on radio resource allocation in cognitive radio sensor networks," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 2, pp. 888–917, 2nd Quart., 2015.
- [22] S. K. Sharma, T. E. Bogale, S. Chatzinotas, B. Ottersten, L. B. Le, and X. Wang, "Cognitive radio techniques under practical imperfections: A survey," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 1858–1884, Jul. 2015.
- [23] S. Joshi and R. K. Mallik, "Cooperative NOMA with AF relaying over Nakagami-m fading in a D2D network," in *Proc. IEEE 89th Veh. Technol. Conf. (VTC-Spring)*, Apr. 2019, pp. 1–6.
- [24] Prnewswire. (2020). Cognitive Radio Market Worth 7.44 Billion USD by 2022. Accessed: Nov. 4, 2020. [Online]. Available: https://www.prnewswire.com/news-releases/cognitive-radio-marketworth-744-billion-usd-by-2022-668809553.html
- [25] G. Bansal and V. Chamola, "Lightweight authentication protocol for inter base station communication in heterogeneous networks," in *Proc. IEEE Conf. Comput. Commun. Workshops (INFOCOM WKSHPS)*, Jul. 2020, pp. 871–876.
- [26] S. Joshi and R. K. Mallik, "Coverage and interference in D2D networks with Poisson cluster process," *IEEE Commun. Lett.*, vol. 22, no. 5, pp. 1098–1101, May 2018.
- [27] S. Joshi and R. K. Mallik, "Analysis of dedicated and shared deviceto-device communication in cellular networks over Nakagami-m fading channels," *IET Commun.*, vol. 11, no. 10, pp. 1600–1609, Jul. 2017.
- [28] M. A. Hossain, R. M. Noor, K.-L.-A. Yau, I. Ahmedy, and S. S. Anjum, "A survey on simultaneous wireless information and power transfer with cooperative relay and future challenges," *IEEE Access*, vol. 7, pp. 19166–19198, 2019.
- [29] S. Buzzi, C.-L. I, T. E. Klein, H. V. Poor, C. Yang, and A. Zappone, "A survey of energy-efficient techniques for 5G networks and challenges ahead," *IEEE J. Sel. Areas Commun.*, vol. 34, no. 4, pp. 697–709, Apr. 2016.
- [30] S. Sudevalayam and P. Kulkarni, "Energy harvesting sensor nodes: Survey and implications," *IEEE Commun. Surveys Tuts.*, vol. 13, no. 3, pp. 443–461, 3rd Quart., 2011.
- [31] R. V. Prasad, S. Devasenapathy, V. S. Rao, and J. Vazifehdan, "Reincarnation in the ambiance: Devices and networks with energy harvesting," *IEEE Commun. Surveys Tuts.*, vol. 16, no. 1, pp. 195–213, 1st Quart., 2014.
- [32] X. Cao, L. Liu, Y. Cheng, and X. Shen, "Towards energy-efficient wireless networking in the big data era: A survey," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 1, pp. 303–332, 1st Quart., 2018.

- [33] Environment + Energy Leader. (2020). Energy Harvesting Promises 4.4 Billion USD Market by 2021. Accessed: Nov. 4, 2020. [Online]. Available: https://www.environmentalleader.com/2011/08/energy-harvestingpromises-4-4bn-market-by-2021/
- [34] N. Zhao, F. R. Yu, and H. Sun, "Adaptive energy-efficient power allocation in green Interference-Alignment-Based wireless networks," *IEEE Trans. Veh. Technol.*, vol. 64, no. 9, pp. 4268–4281, Sep. 2015.
- [35] R. Xie, F. R. Yu, H. Ji, and Y. Li, "Energy-efficient resource allocation for heterogeneous cognitive radio networks with femtocells," *IEEE Trans. Wireless Commun.*, vol. 11, no. 11, pp. 3910–3920, Nov. 2012.
- [36] Sallan. (2020). Revolutionizing Data Center Energy Efficiency. Accessed: Nov. 4, 2020. [Online]. Available: https://sallan.org/pdfdocs/McKinsey_Data_Center_Efficiency.pdf
- [37] D. Gunduz, K. Stamatiou, N. Michelusi, and M. Zorzi, "Designing intelligent energy harvesting communication systems," *IEEE Commun. Mag.*, vol. 52, no. 1, pp. 210–216, Jan. 2014.
- [38] V. Chamola, P. Narang, and B. Sikdar, "Downlink power control for latency aware grid energy savings in green cellular networks," in *Proc. IEEE 37th Sarnoff Symp.*, Sep. 2016, pp. 1–6.
- [39] C. Despins, F. Labeau, T. Le Ngoc, R. Labelle, M. Cheriet, C. Thibeault, F. Gagnon, A. Leon-Garcia, O. Cherkaoui, B. St. Arnaud, J. Mcneill, Y. Lemieux, and M. Lemay, "Leveraging green communications for carbon emission reductions: Techniques, testbeds, and emerging carbon footprint standards," *IEEE Commun. Mag.*, vol. 49, no. 8, pp. 101–109, Aug. 2011.
- [40] V. Chamola and B. Sikdar, "Power outage estimation and resource dimensioning for solar powered cellular base stations," *IEEE Trans. Commun.*, vol. 64, no. 12, pp. 5278–5289, Dec. 2016.
- [41] V. Chamola and B. Sikdar, "A multistate Markov model for dimensioning solar powered cellular base stations," *IEEE Trans. Sustain. Energy*, vol. 6, no. 4, pp. 1650–1652, Oct. 2015.
- [42] Wikipedia Contributors. Energy Harvesting. Accessed: Nov. 4, 2020. [Online]. Available: https://en.wikipedia.org/wiki/Energy_harvesting
- [43] The Conversation. (2020). Creative Destruction: The COVID-19 Economic Crisis is Accelerating the Demise of Fossil Fuels. Accessed: Nov. 4, 2020. [Online]. Available: https://theconversation.com/ creative-destruction-the-covid-19-economic-crisis-is-accelerating-thedemise-of-fossil-fuels-143739
- [44] N. Michelusi, L. Badia, and M. Zorzi, "Optimal transmission policies for energy harvesting devices with limited state-of-charge knowledge," *IEEE Trans. Commun.*, vol. 62, no. 11, pp. 3969–3982, Nov. 2014.
- [45] I. Krikidis, G. Zheng, and B. Ottersten, "Harvest-use cooperative networks with half/full-duplex relaying," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, Apr. 2013, pp. 4256–4260.
- [46] X. Kang, Y.-K. Chia, C. K. Ho, and S. Sun, "Cost minimization for fading channels with energy harvesting and conventional energy," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2014, pp. 1910–1915.
- [47] R. Rajesh, V. Sharma, and P. Viswanath, "Capacity of Gaussian channels with energy harvesting and processing cost," *IEEE Trans. Inf. Theory*, vol. 60, no. 5, pp. 2563–2575, May 2014.
- [48] H. Chen, Y. Li, J. L. Rebelatto, B. F. Uchoa-Filho, and B. Vucetic, "Harvest-then-cooperate: Wireless-powered cooperative communications," *IEEE Trans. Signal Process.*, vol. 63, no. 7, pp. 1700–1711, Apr. 2015.
- [49] B. Devillers and D. Gündüz, "A general framework for the optimization of energy harvesting communication systems with battery imperfections," J. Commun. Netw., vol. 14, no. 2, pp. 130–139, Apr. 2012.
- [50] P. S. Khairnar and N. B. Mehta, "Discrete-rate adaptation and selection in energy harvesting wireless systems," *IEEE Trans. Wireless Commun.*, vol. 14, no. 1, pp. 219–229, Jan. 2015.
- [51] J. Xu and R. Zhang, "Throughput optimal policies for energy harvesting wireless transmitters with non-ideal circuit power," *IEEE J. Sel. Areas Commun.*, vol. 32, no. 2, pp. 322–332, Feb. 2014.
- [52] K. Tutuncuoglu, A. Yener, and S. Ulukus, "Optimum policies for an energy harvesting transmitter under energy storage losses," *IEEE J. Sel. Areas Commun.*, vol. 33, no. 3, pp. 467–481, Mar. 2015.
- [53] V. Chamola and B. Sikdar, "Solar powered cellular base stations: Current scenario, issues and proposed solutions," *IEEE Commun. Mag.*, vol. 54, no. 5, pp. 108–114, May 2016.
- [54] V. Chamola and B. Sikdar, "Outage estimation for solar powered cellular base stations," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2015, pp. 172–177.

- [55] V. Chamola and B. Sikdar, "Resource provisioning and dimensioning for solar powered cellular base stations," in *Proc. IEEE Global Commun. Conf.*, Dec. 2014, pp. 2498–2503.
- [56] V. Chamola and B. Sikdar, "Dimensioning stand-alone cellular base station using series-of-worst-months meteorological data," in *Proc. IEEE Int. Conf. Commun. Syst.*, Nov. 2014, pp. 339–343.
- [57] C. Chen, G. Wang, P. D. Diamantoulakis, R. He, G. K. Karagiannidis, and C. Tellambura, "Signal detection and optimal antenna selection for ambient backscatter communications with multi-antenna tags," *IEEE Trans. Commun.*, vol. 68, no. 1, pp. 466–479, Jan. 2020.
- [58] H. Guo, Y.-C. Liang, R. Long, S. Xiao, and Q. Zhang, "Resource allocation for symbiotic radio system with fading channels," *IEEE Access*, vol. 7, pp. 34333–34347, 2019.
- [59] X. Kang, Y.-C. Liang, and J. Yang, "Riding on the primary: A new spectrum sharing paradigm for wireless-powered IoT devices," in *Proc. IEEE Int. Conf. Commun. (ICC)*, May 2017, pp. 1–6.
- [60] H. Guo, Y.-C. Liang, R. Long, and Q. Zhang, "Cooperative ambient backscatter system: A symbiotic radio paradigm for passive IoT," *IEEE Wireless Commun. Lett.*, vol. 8, no. 4, pp. 1191–1194, Aug. 2019.
- [61] R. Kishore, S. Gurugopinath, P. C. Sofotasios, S. Muhaidat, and N. Al-Dhahir, "Opportunistic ambient backscatter communication in RFpowered cognitive radio networks," *IEEE Trans. Cognit. Commun. Netw.*, vol. 5, no. 2, pp. 413–426, Jun. 2019.
- [62] T. D. Ponnimbaduge Perera, D. N. K. Jayakody, S. K. Sharma, S. Chatzinotas, and J. Li, "Simultaneous wireless information and power transfer (SWIPT): Recent advances and future challenges," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 1, pp. 264–302, 1st Quart., 2018.
- [63] R. Zhang and C. K. Ho, "MIMO broadcasting for simultaneous wireless information and power transfer," *IEEE Trans. Wireless Commun.*, vol. 12, no. 5, pp. 1989–2001, May 2013.
- [64] X. Lu, P. Wang, D. Niyato, and Z. Han, "Resource allocation in wireless networks with RF energy harvesting and transfer," *IEEE Netw.*, vol. 29, no. 6, pp. 68–75, Nov. 2015.
- [65] I. Krikidis, S. Timotheou, S. Nikolaou, G. Zheng, D. W. K. Ng, and R. Schober, "Simultaneous wireless information and power transfer in modern communication systems," *IEEE Commun. Mag.*, vol. 52, no. 11, pp. 104–110, Nov. 2014.
- [66] H. Chen, Y. Li, Y. Jiang, Y. Ma, and B. Vucetic, "Distributed power splitting for SWIPT in relay interference channels using game theory," *IEEE Trans. Wireless Commun.*, vol. 14, no. 1, pp. 410–420, Jan. 2015.
- [67] Y. Xu, C. Shen, Z. Ding, X. Sun, S. Yan, G. Zhu, and Z. Zhong, "Joint beamforming and power-splitting control in downlink cooperative SWIPT NOMA systems," *IEEE Trans. Signal Process.*, vol. 65, no. 18, pp. 4874–4886, Sep. 2017.
- [68] Y. Gu and S. Aissa, "RF-based energy harvesting in decode-and-forward relaying systems: Ergodic and outage capacities," *IEEE Trans. Wireless Commun.*, vol. 14, no. 11, pp. 6425–6434, Nov. 2015.
- [69] A. A. Nasir, X. Zhou, S. Durrani, and R. A. Kennedy, "Wireless-powered relays in cooperative communications: Time-switching relaying protocols and throughput analysis," *IEEE Trans. Commun.*, vol. 63, no. 5, pp. 1607–1622, May 2015.
- [70] F. Zhao, L. Wei, and H. Chen, "Optimal time allocation for wireless information and power transfer in wireless powered communication systems," *IEEE Trans. Veh. Technol.*, vol. 65, no. 3, pp. 1830–1835, Mar. 2016.
- [71] E. Boshkovska, D. W. K. Ng, N. Zlatanov, and R. Schober, "Practical nonlinear energy harvesting model and resource allocation for SWIPT systems," *IEEE Commun. Lett.*, vol. 19, no. 12, pp. 2082–2085, Dec. 2015.
- [72] L. Lei, Y. Kuang, X. S. Shen, K. Yang, J. Qiao, and Z. Zhong, "Optimal reliability in energy harvesting industrial wireless sensor networks," *IEEE Trans. Wireless Commun.*, vol. 15, no. 8, pp. 5399–5413, Aug. 2016.
- [73] C. Huang, R. Zhang, and S. Cui, "Optimal power allocation for outage probability minimization in fading channels with energy harvesting constraints," *IEEE Trans. Wireless Commun.*, vol. 13, no. 2, pp. 1074–1087, Feb. 2014.
- [74] S. Wei, W. Guan, and K. J. R. Liu, "Power scheduling for energy harvesting wireless communications with battery capacity constraint," *IEEE Trans. Wireless Commun.*, vol. 14, no. 8, pp. 4640–4653, Aug. 2015.
- [75] N. Roseveare and B. Natarajan, "An alternative perspective on utility maximization in energy-harvesting wireless sensor networks," *IEEE Trans. Veh. Technol.*, vol. 63, no. 1, pp. 344–356, Jan. 2014.
- [76] R. Srivastava and C. E. Koksal, "Basic performance limits and tradeoffs in energy-harvesting sensor nodes with finite data and energy storage," *IEEE/ACM Trans. Netw.*, vol. 21, no. 4, pp. 1049–1062, Aug. 2013.

- [77] Y. Wei, F. R. Yu, M. Song, and Z. Han, "User scheduling and resource allocation in HetNets with hybrid energy supply: An actor-critic reinforcement learning approach," *IEEE Trans. Wireless Commun.*, vol. 17, no. 1, pp. 680–692, Jan. 2018.
- [78] Z. Wang, V. Aggarwal, and X. Wang, "Power allocation for energy harvesting transmitter with causal information," *IEEE Trans. Commun.*, vol. 62, no. 11, pp. 4080–4093, Nov. 2014.
- [79] X. Cao, X. Zhou, L. Liu, and Y. Cheng, "Energy-efficient spectrum sensing for cognitive radio enabled remote state estimation over wireless channels," *IEEE Trans. Wireless Commun.*, vol. 14, no. 4, pp. 2058–2071, Apr. 2015.
- [80] K. J. Prabuchandran, S. K. Meena, and S. Bhatnagar, "Q-learning based energy management policies for a single sensor node with finite buffer," *IEEE Wireless Commun. Lett.*, vol. 2, no. 1, pp. 82–85, Feb. 2013.
- [81] S. Ulukus, A. Yener, E. Erkip, O. Simeone, M. Zorzi, P. Grover, and K. Huang, "Energy harvesting wireless communications: A review of recent advances," *IEEE J. Sel. Areas Commun.*, vol. 33, no. 3, pp. 360–381, Jan. 2015, doi: 10.1109/JSAC.2015.2391531.
- [82] V. L. Quintero, C. Estevez, M. E. Orchard, and A. Perez, "Improvements of energy-efficient techniques in WSNs: A MAC-protocol approach," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 2, pp. 1188–1208, 2nd Quart., 2019.
- [83] T. A. Al-Janabi and H. S. Al-Raweshidy, "An energy efficient hybrid MAC protocol with dynamic sleep-based scheduling for high density IoT networks," *IEEE Internet Things J.*, vol. 6, no. 2, pp. 2273–2287, Apr. 2019.
- [84] X. Lu, P. Wang, D. Niyato, D. I. Kim, and Z. Han, "Wireless networks with RF energy harvesting: A contemporary survey," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 2, pp. 757–789, 2nd Quart., 2015.
- [85] D. Ma, G. Lan, M. Hassan, W. Hu, and S. K. Das, "Sensing, computing, and communications for energy harvesting IoTs: A survey," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 2, pp. 1222–1250, 2nd Quart., 2020.
- [86] A. A. Khan, M. H. Rehmani, and A. Rachedi, "Cognitive-radio-based Internet of Things: Applications, architectures, spectrum related functionalities, and future research directions," *IEEE Wireless Commun.*, vol. 24, no. 3, pp. 17–25, Jun. 2017.
- [87] A. Ali and W. Hamouda, "Advances on spectrum sensing for cognitive radio networks: Theory and applications," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 2, pp. 1277–1304, 2nd Quart., 2017.
- [88] R. Han, Y. Gao, C. Wu, and D. Lu, "An effective multi-objective optimization algorithm for spectrum allocations in the cognitive-radio-based Internet of Things," *IEEE Access*, vol. 6, pp. 12858–12867, 2018.
- [89] L. Zhang, Z. Cai, P. Li, L. Wang, and X. Wang, "Spectrum-availability based routing for cognitive sensor networks," *IEEE Access*, vol. 5, pp. 4448–4457, 2017.
- [90] A. Taherpour, H. Mokhtarzadeh, and T. Khattab, "Optimized error probability for weighted collaborative spectrum sensing in time- and energylimited cognitive radio networks," *IEEE Trans. Veh. Technol.*, vol. 66, no. 10, pp. 9035–9049, Oct. 2017.
- [91] P. V. Tuan and I. Koo, "Throughput maximisation by optimising detection thresholds in full-duplex cognitive radio networks," *IET Commun.*, vol. 10, no. 11, pp. 1355–1364, Jul. 2016.
- [92] O. P. Awe, A. Deligiannis, and S. Lambotharan, "Spatio-temporal spectrum sensing in cognitive radio networks using beamformer-aided SVM algorithms," *IEEE Access*, vol. 6, pp. 25377–25388, 2018.
- [93] C. Politis, S. Maleki, C. G. Tsinos, K. P. Liolis, S. Chatzinotas, and B. Ottersten, "Simultaneous sensing and transmission for cognitive radios with imperfect signal cancellation," *IEEE Trans. Wireless Commun.*, vol. 16, no. 9, pp. 5599–5615, Sep. 2017.
- [94] S. MacDonald, D. C. Popescu, and O. Popescu, "Analyzing the performance of spectrum sensing in cognitive radio systems with dynamic PU activity," *IEEE Commun. Lett.*, vol. 21, no. 9, pp. 2037–2040, Sep. 2017.
- [95] X. Liu, B. G. Evans, and K. Moessner, "Energy-efficient sensor scheduling algorithm in cognitive radio networks employing heterogeneous sensors," *IEEE Trans. Veh. Technol.*, vol. 64, no. 3, pp. 1243–1249, Mar. 2015.
- [96] W. Cheng, X. Zhang, and H. Zhang, "Full-duplex spectrum-sensing and MAC-protocol for multichannel nontime-slotted cognitive radio networks," *IEEE J. Sel. Areas Commun.*, vol. 33, no. 5, pp. 820–831, May 2015.
- [97] T. Yucek and H. Arslan, "A survey of spectrum sensing algorithms for cognitive radio applications," *IEEE Commun. Surveys Tuts.*, vol. 11, no. 1, pp. 116–130, 1st Quart., 2009.

- [98] S. Shankar N., C. Cordeiro, and K. Challapali, "Spectrum agile radios: Utilization and sensing architectures," in *Proc. 1st IEEE Int. Symp. New Frontiers Dyn. Spectr. Access Netw. (DySPAN)*, Nov. 2005, pp. 160–169.
- [99] H. Chen, L. Liu, J. Matyjas, and M. Medley, "Optimal resource allocation for sensing based spectrum sharing cognitive radio networks," in *Proc. IEEE Global Commun. Conf.*, Dec. 2014, pp. 899–904.
- [100] L. De Vito, "A review of wideband spectrum sensing methods for cognitive radios," in *Proc. IEEE Int. Instrum. Meas. Technol. Conf.*, May 2012, pp. 2257–2262.
- [101] I. F. Akyildiz, W.-Y. Lee, M. C. Vuran, and S. Mohanty, "NeXt generation/dynamic spectrum access/cognitive radio wireless networks: A survey," *Comput. Netw.*, vol. 50, no. 13, pp. 2127–2159, Sep. 2006.
- [102] C. Liu, M. Li, and M.-L. Jin, "Blind energy-based detection for spatial spectrum sensing," *IEEE Wireless Commun. Lett.*, vol. 4, no. 1, pp. 98–101, Feb. 2015.
- [103] A. Margoosian, J. Abouei, and K. N. Plataniotis, "An accurate kernelized energy detection in Gaussian and non-Gaussian/impulsive noises," *IEEE Trans. Signal Process.*, vol. 63, no. 21, pp. 5621–5636, Nov. 2015.
- [104] I. Sobron, P. S. R. Diniz, W. A. Martins, and M. Velez, "Energy detection technique for adaptive spectrum sensing," *IEEE Trans. Commun.*, vol. 63, no. 3, pp. 617–627, Mar. 2015.
- [105] O. Semiari, B. Maham, and C. Yuen, "On the effect of I/Q imbalance on energy detection and a novel four-level hypothesis spectrum sensing," *IEEE Trans. Veh. Technol.*, vol. 63, no. 8, pp. 4136–4141, Oct. 2014.
- [106] F. Salahdine, H. E. Ghazi, N. Kaabouch, and W. F. Fihri, "Matched filter detection with dynamic threshold for cognitive radio networks," in *Proc. Int. Conf. Wireless Netw. Mobile Commun. (WINCOM)*, Oct. 2015, pp. 1–6.
- [107] X. Zhang, R. Chai, and F. Gao, "Matched filter based spectrum sensing and power level detection for cognitive radio network," in *Proc. IEEE Global Conf. Signal Inf. Process. (GlobalSIP)*, Dec. 2014, pp. 1267–1270.
- [108] Q. Lv and F. Gao, "Matched filter based spectrum sensing and power level recognition with multiple antennas," in *Proc. IEEE China Summit Int. Conf. Signal Inf. Process. (ChinaSIP)*, Jul. 2015, pp. 305–309.
- [109] A. Nasser, A. Mansour, K. C. Yao, H. Charara, and M. Chaitou, "Efficient spectrum sensing approaches based on waveform detection," in *Proc. 3rd Int. Conf. e-Technol. Netw. for Develop. (ICeND)*, Apr. 2014, pp. 13–17.
- [110] N. A. El-Alfi, H. M. Abdel-Atty, and M. A. Mohamed, "Sub-Nyquist cyclostationary detection of GFDM for wideband spectrum sensing," *IEEE Access*, vol. 7, pp. 86403–86411, 2019.
- [111] A. Tani, R. Fantacci, and D. Marabissi, "A low-complexity cyclostationary spectrum sensing for interference avoidance in femtocell LTE-A-based networks," *IEEE Trans. Veh. Technol.*, vol. 65, no. 4, pp. 2747–2753, Apr. 2016.
- [112] P. Sepidband and K. Entesari, "A CMOS spectrum sensor based on quasi-cyclostationary feature detection for cognitive radios," *IEEE Trans. Microw. Theory Techn.*, vol. 63, no. 12, pp. 4098–4109, Dec. 2015.
- [113] A. Kortun, T. Ratnarajah, M. Sellathurai, Y.-C. Liang, and Y. Zeng, "On the eigenvalue-based spectrum sensing and secondary user throughput," *IEEE Trans. Veh. Technol.*, vol. 63, no. 3, pp. 1480–1486, Mar. 2014.
- [114] S. K. Sharma, S. Chatzinotas, and B. Ottersten, "Eigenvalue-based sensing and SNR estimation for cognitive radio in presence of noise correlation," *IEEE Trans. Veh. Technol.*, vol. 62, no. 8, pp. 3671–3684, Oct. 2013.
- [115] H. Sun, A. Nallanathan, C.-X. Wang, and Y. Chen, "Wideband spectrum sensing for cognitive radio networks: A survey," *IEEE Wireless Commun.*, vol. 20, no. 2, pp. 74–81, Apr. 2013.
- [116] Z. Quan, S. Cui, A. H. Sayed, and H. V. Poor, "Optimal multiband joint detection for spectrum sensing in cognitive radio networks," *IEEE Trans. Signal Process.*, vol. 57, no. 3, pp. 1128–1140, Mar. 2009.
- [117] J. F. M. Lee, A. López-Parrado, and J. Velasco-Medina, "Hardware implementation of FFT-based spectrum sensing techniques for cognitive radio," in *Proc. IEEE 7th Latin Amer. Symp. Circuits Syst. (LASCAS)*, Feb. 2016, pp. 123–126.
- [118] Y. Feng and X. Wang, "Adaptive multiband spectrum sensing," *IEEE Wireless Commun. Lett.*, vol. 1, no. 2, pp. 121–124, Apr. 2012.
- [119] W. Ejaz and M. Ibnkahla, "Multiband spectrum sensing and resource allocation for IoT in cognitive 5G networks," *IEEE Internet Things J.*, vol. 5, no. 1, pp. 150–163, Feb. 2018.
- [120] H. Hosseini, A. Anpalagan, S. Habib, K. Raahemifar, and S. Erkucuk, "Joint wavelet-based spectrum sensing and FBMC modulation for cognitive mmWave small cell networks," *IET Commun.*, vol. 10, no. 14, pp. 1803–1809, Sep. 2016.

- [121] D. Capriglione, G. Cerro, L. Ferrigno, and G. Miele, "Analysis and implementation of a wavelet based spectrum sensing method for low SNR scenarios," in *Proc. IEEE 17th Int. Symp. World Wireless, Mobile Multimedia Netw. (WoWMoM)*, Jun. 2016, pp. 1–6.
- [122] A. Kumar, S. Saha, and R. Bhattacharya, "Improved wavelet transform based edge detection for wide band spectrum sensing in cognitive radio," in *Proc. USNC-URSI Radio Sci. Meeting*, Jun. 2016, pp. 21–22.
- [123] S. Dikmese, M. Renfors, and I. Guvenc, "Flexible filter bank based spectrum sensing and waveform processing for mission critical communications," in *Proc. IEEE Mil. Commun. Conf. (MILCOM)*, Oct. 2017, pp. 174–179.
- [124] I. Raghu, S. S. Chowdary, and E. Elias, "Efficient spectrum sensing for cognitive radio using cosine modulated filter banks," in *Proc. IEEE Region Conf. (TENCON)*, Nov. 2016, pp. 2086–2089.
- [125] M. Kim and J.-I. Takada, "Efficient multi-channel wideband spectrum sensing technique using filter bank," in *Proc. IEEE 20th Int. Symp. Pers.*, *Indoor Mobile Radio Commun.*, Sep. 2009, pp. 1014–1018.
- [126] M. Lin and A. P. Vinod, "Progressive decimation filter banks for variable resolution spectrum sensing in cognitive radios," in *Proc. 17th Int. Conf. Telecommun.*, 2010, pp. 857–863.
- [127] B. Farhang-Boroujeny, "Filter bank spectrum sensing for cognitive radios," *IEEE Trans. Signal Process.*, vol. 56, no. 5, pp. 1801–1811, May 2008.
- [128] Z. Qin, J. Wang, J. Chen, and L. Wang, "Adaptive compressed spectrum sensing based on cross validation in WideBand cognitive radio system," *IEEE Syst. J.*, vol. 11, no. 4, pp. 2422–2431, Dec. 2017.
- [129] Z. Qin, Y. Gao, and C. G. Parini, "Data-assisted low complexity compressive spectrum sensing on real-time signals under sub-Nyquist rate," *IEEE Trans. Wireless Commun.*, vol. 15, no. 2, pp. 1174–1185, Feb. 2016.
- [130] Z. Li, B. Chang, S. Wang, A. Liu, F. Zeng, and G. Luo, "Dynamic compressive wide-band spectrum sensing based on channel energy reconstruction in cognitive Internet of Things," *IEEE Trans. Ind. Informat.*, vol. 14, no. 6, pp. 2598–2607, Jun. 2018.
- [131] F. Salahdine, N. Kaabouch, and H. E. Ghazi, "Bayesian compressive sensing with circulant matrix for spectrum sensing in cognitive radio networks," in *Proc. IEEE 7th Annu. Ubiquitous Comput., Electron. Mobile Commun. Conf. (UEMCON)*, Oct. 2016, pp. 1–6.
- [132] X. Zhang, Y. Ma, H. Qi, Y. Gao, Z. Xie, Z. Xie, M. Zhang, X. Wang, G. Wei, and Z. Li, "Distributed compressive sensing augmented wideband spectrum sharing for cognitive IoT," *IEEE Internet Things J.*, vol. 5, no. 4, pp. 3234–3245, Aug. 2018.
- [133] B. Khalfi, B. Hamdaoui, M. Guizani, and N. Zorba, "Efficient spectrum availability information recovery for wideband DSA networks: A weighted compressive sampling approach," *IEEE Trans. Wireless Commun.*, vol. 17, no. 4, pp. 2162–2172, Apr. 2018.
- [134] J. Jiang, H. Sun, D. Baglee, and H. Vincent Poor, "Achieving autonomous compressive spectrum sensing for cognitive radios," *IEEE Trans. Veh. Technol.*, vol. 65, no. 3, pp. 1281–1291, Mar. 2016.
- [135] Z. Qin, Y. Liu, Y. Gao, M. Elkashlan, and A. Nallanathan, "Wireless powered cognitive radio networks with compressive sensing and matrix completion," *IEEE Trans. Commun.*, vol. 65, no. 4, pp. 1464–1476, Apr. 2017.
- [136] M. M. Mabrook, G. A. Fahmy, A. I. Hussein, and M. A. Abdelghany, "Adaptive blind wideband spectrum sensing for cognitive radio based on sub-Nyquist sampling technique," in *Proc. 28th Int. Conf. Microelectron.* (*ICM*), Dec. 2016, pp. 141–144.
- [137] J. Dong, H. Li, Z. Fan, Q. Zhou, and S. Hua, "A joint optimization algorithm using adaptive minimum coset number based discrete multicoset sampling," *IEEE Access*, vol. 8, pp. 168659–168670, 2020.
- [138] R. Venkataramani and Y. Bresler, "Perfect reconstruction formulas and bounds on aliasing error in sub-nyquist nonuniform sampling of multiband signals," *IEEE Trans. Inf. Theory*, vol. 46, no. 6, pp. 2173–2183, Sep. 2000.
- [139] P. Feng and Y. Bresler, "Spectrum-blind minimum-rate sampling and reconstruction of multiband signals," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process. Conf.*, vol. 3, May 1996, pp. 1688–1691.
- [140] S. Maleki and G. Leus, "Censored truncated sequential spectrum sensing for cognitive radio networks," in *Proc. 17th Int. Conf. Digit. Signal Process. (DSP)*, Jul. 2011, pp. 1–8.
- [141] D. M. Martínez and A. G. Andrade, "On the reduction of the noise uncertainty effects in energy detection for spectrum sensing in cognitive radios," in *Proc. IEEE 25th Annu. Int. Symp. Pers., Indoor, Mobile Radio Commun. (PIMRC)*, Sep. 2014, pp. 1975–1979.

- [142] K. Cichoń, A. Kliks, and H. Bogucka, "Energy-efficient cooperative spectrum sensing: A survey," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 3, pp. 1861–1886, 3rd Quart., 2016.
- [143] W. Na, J. Yoon, S. Cho, D. Griffith, and N. Golmie, "Centralized cooperative directional spectrum sensing for cognitive radio networks," *IEEE Trans. Mobile Comput.*, vol. 17, no. 6, pp. 1260–1274, Jun. 2018.
- [144] L. Khalid and A. Anpalagan, "Reliability-based decision fusion scheme for cooperative spectrum sensing," *IET Commun.*, vol. 8, no. 14, pp. 2423–2432, Sep. 2014.
- [145] Y. Wang, H. Li, and L. Qian, "Belief propagation and quickest detection-based cooperative spectrum sensing in heterogeneous and dynamic environments," *IEEE Trans. Wireless Commun.*, vol. 16, no. 11, pp. 7446–7459, Nov. 2017.
- [146] L. Ma, Y. Xiang, Q. Pei, Y. Xiang, and H. Zhu, "Robust reputation-based cooperative spectrum sensing via imperfect common control channel," *IEEE Trans. Veh. Technol.*, vol. 67, no. 5, pp. 3950–3963, May 2018.
- [147] A. Zaeemzadeh, M. Joneidi, N. Rahnavard, and G.-J. Qi, "Co-SpOT: Cooperative spectrum opportunity detection using Bayesian clustering in spectrum-heterogeneous cognitive radio networks," *IEEE Trans. Cognit. Commun. Netw.*, vol. 4, no. 2, pp. 206–219, Jun. 2018.
- [148] M. Sun, M. Jin, Q. Guo, and Y. Li, "Cooperative spectrum sensing under ambient malicious interferences," *IEEE Commun. Lett.*, vol. 22, no. 2, pp. 432–435, Feb. 2018.
- [149] H. Lin, J. Hu, J. Ma, L. Xu, and Z. Yu, "A secure collaborative spectrum sensing strategy in cyber-physical systems," *IEEE Access*, vol. 5, pp. 27679–27690, 2017.
- [150] W. Zhang, Y. Yang, and C. K. Yeo, "Cluster-based cooperative spectrum sensing assignment strategy for heterogeneous cognitive radio network," *IEEE Trans. Veh. Technol.*, vol. 64, no. 6, pp. 2637–2647, Jun. 2015.
- [151] A. Celik and A. E. Kamal, "Multi-objective clustering optimization for multi-channel cooperative spectrum sensing in heterogeneous green CRNs," *IEEE Trans. Cognit. Commun. Netw.*, vol. 2, no. 2, pp. 150–161, Jun. 2016.
- [152] Y. Jiao, I. Joe, and P. Yin, "Clustering scheme for cooperative spectrum sensing in cognitive radio networks," *IET Commun.*, vol. 10, no. 13, pp. 1590–1595, Sep. 2016.
- [153] M. Zheng, L. Chen, W. Liang, H. Yu, and J. Wu, "Energy-efficiency maximization for cooperative spectrum sensing in cognitive sensor networks," *IEEE Trans. Green Commun. Netw.*, vol. 1, no. 1, pp. 29–39, Mar. 2017.
- [154] S. Zhang, Q. Wu, S. Xu, and G. Y. Li, "Fundamental green tradeoffs: Progresses, challenges, and impacts on 5G networks," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 1, pp. 33–56, 1st Quart., 2017.
- [155] Z. Hasan, H. Boostanimehr, and V. K. Bhargava, "Green cellular networks: A survey, some research issues and challenges," *IEEE Commun. Surveys Tuts.*, vol. 13, no. 4, pp. 524–540, Nov. 2011.
- [156] M. Sami, N. K. Noordin, and M. Khabazian, "A TDMA-based cooperative MAC protocol for cognitive networks with opportunistic energy harvesting," *IEEE Commun. Lett.*, vol. 20, no. 4, pp. 808–811, Apr. 2016.
- [157] Y. Chen, S. Zhang, S. Xu, and G. Li, "Fundamental trade-offs on green wireless networks," *IEEE Commun. Mag.*, vol. 49, no. 6, pp. 30–37, Jun. 2011.
- [158] V. Chamola, B. Sikdar, and B. Krishnamachari, "Delay aware resource management for grid energy savings in green cellular base stations with hybrid power supplies," *IEEE Trans. Commun.*, vol. 65, no. 3, pp. 1092–1104, Mar. 2017.
- [159] V. Chamola, B. Krishnamachari, and B. Sikdar, "Green energy and delay aware downlink power control and user association for off-grid solarpowered base stations," *IEEE Syst. J.*, vol. 12, no. 3, pp. 2622–2633, Sep. 2018.
- [160] W. Ejaz, M. Naeem, A. Shahid, A. Anpalagan, and M. Jo, "Efficient energy management for the Internet of Things in smart cities," *IEEE Commun. Mag.*, vol. 55, no. 1, pp. 84–91, Jan. 2017.
- [161] D. Zhang, Z. Chen, J. Ren, N. Zhang, M. K. Awad, H. Zhou, and X. S. Shen, "Energy-harvesting-aided spectrum sensing and data transmission in heterogeneous cognitive radio sensor network," *IEEE Trans. Veh. Technol.*, vol. 66, no. 1, pp. 831–843, Jan. 2017.
- [162] E. Pei, L. Li, W. Cheng, and J. Jiang, "An energy efficient spectrum sensing algorithm in EH-HCRSN," *IEEE Access*, vol. 6, pp. 63021–63032, 2018.
- [163] A. Celik and A. E. Kamal, "Green cooperative spectrum sensing and scheduling in heterogeneous cognitive radio networks," *IEEE Trans. Cognit. Commun. Netw.*, vol. 2, no. 3, pp. 238–248, Sep. 2016.
- VOLUME 9, 2021

- [164] X. Ding, Y. Zou, G. Zhang, X. Chen, X. Wang, and L. Hanzo, "The security-reliability tradeoff of multiuser scheduling-aided energy harvesting cognitive radio networks," *IEEE Trans. Commun.*, vol. 67, no. 6, pp. 3890–3904, Jun. 2019.
- [165] X. Gao, P. Wang, D. Niyato, K. Yang, and J. An, "Auction-based time scheduling for backscatter-aided RF-powered cognitive radio networks," *IEEE Trans. Wireless Commun.*, vol. 18, no. 3, pp. 1684–1697, Mar. 2019.
- [166] Z. Zhou, C. Zhang, J. Wang, B. Gu, S. Mumtaz, J. Rodriguez, and X. Zhao, "Energy-efficient resource allocation for energy harvestingbased cognitive machine-to-machine communications," *IEEE Trans. Cognit. Commun. Netw.*, vol. 5, no. 3, pp. 595–607, Sep. 2019.
- [167] Y. Wang, Y. Wang, F. Zhou, Y. Wu, and H. Zhou, "Resource allocation in wireless powered cognitive radio networks based on a practical nonlinear energy harvesting model," *IEEE Access*, vol. 5, pp. 17618–17626, 2017.
- [168] Z. Hu, N. Wei, and Z. Zhang, "Optimal resource allocation for harvested energy maximization in wideband cognitive radio network with SWIPT," *IEEE Access*, vol. 5, pp. 23383–23394, 2017.
- [169] D. Xu and Q. Li, "Cooperative resource allocation in cognitive radio networks with wireless powered primary users," *IEEE Wireless Commun. Lett.*, vol. 6, no. 5, pp. 658–661, Oct. 2017.
- [170] A. Bhowmick, S. D. Roy, and S. Kundu, "Throughput of a cognitive radio network with energy-harvesting based on primary user signal," *IEEE Wireless Commun. Lett.*, vol. 5, no. 2, pp. 136–139, Apr. 2016.
- [171] F. Zhou, N. C. Beaulieu, J. Cheng, Z. Chu, and Y. Wang, "Robust maxmin fairness resource allocation in sensing-based wideband cognitive radio with SWIPT: Imperfect channel sensing," *IEEE Syst. J.*, vol. 12, no. 3, pp. 2361–2372, Sep. 2018.
- [172] D. Zhang, Z. Chen, M. K. Awad, N. Zhang, H. Zhou, and X. S. Shen, "Utility-optimal resource management and allocation algorithm for energy harvesting cognitive radio sensor networks," *IEEE J. Sel. Areas Commun.*, vol. 34, no. 12, pp. 3552–3565, Dec. 2016.
- [173] S. Yin, Z. Qu, and S. Li, "Achievable throughput optimization in energy harvesting cognitive radio systems," *IEEE J. Sel. Areas Commun.*, vol. 33, no. 3, pp. 407–422, Mar. 2015.
- [174] M. Usman and I. Koo, "Access strategy for hybrid underlay-overlay cognitive radios with energy harvesting," *IEEE Sensors J.*, vol. 14, no. 9, pp. 3164–3173, Sep. 2014.
- [175] J. A. Ansere, G. Han, H. Wang, C. Choi, and C. Wu, "A reliable energy efficient dynamic spectrum sensing for cognitive radio IoT networks," *IEEE Internet Things J.*, vol. 6, no. 4, pp. 6748–6759, Aug. 2019.
- [176] L. Lv, L. Yang, H. Jiang, T. H. Luan, and J. Chen, "When NOMA meets multiuser cognitive radio: Opportunistic cooperation and user scheduling," *IEEE Trans. Veh. Technol.*, vol. 67, no. 7, pp. 6679–6684, Jul. 2018.
- [177] O. A. Amodu, M. Othman, N. K. Noordin, and I. Ahmad, "Outage analysis of energy-harvesting-based relay-assisted random underlay cognitive radio networks with multihop primary transmissions," *IEEE Syst. J.*, early access, Aug. 7, 2020, doi: 10.1109/JSYST.2020.3011040.
- [178] T. Li, P. Fan, and K. B. Letaief, "Outage probability of energy harvesting relay-aided cooperative networks over Rayleigh fading channel," *IEEE Trans. Veh. Technol.*, vol. 65, no. 2, pp. 972–978, Feb. 2016.
- [179] S. S. Kalamkar and A. Banerjee, "Interference-aided energy harvesting: Cognitive relaying with multiple primary transceivers," *IEEE Trans. Cognit. Commun. Netw.*, vol. 3, no. 3, pp. 313–327, Sep. 2017.
- [180] D. S. Gurjar, H. H. Nguyen, and H. D. Tuan, "Wireless information and power transfer for IoT applications in overlay cognitive radio networks," *IEEE Internet Things J.*, vol. 6, no. 2, pp. 3257–3270, Apr. 2019.
- [181] G. Han, J.-K. Zhang, and X. Mu, "Joint optimization of energy harvesting and detection threshold for energy harvesting cognitive radio networks," *IEEE Access*, vol. 4, pp. 7212–7222, 2016.
- [182] M. Zheng, C. Xu, H. Yu, and W. Liang, "Harvesting-throughput tradeoff for RF-powered underlay cognitive radio networks," *Electron. Lett.*, vol. 52, no. 10, pp. 881–883, May 2016.
- [183] S. Yin, E. Zhang, Z. Qu, L. Yin, and S. Li, "Optimal cooperation strategy in cognitive radio systems with energy harvesting," *IEEE Trans. Wireless Commun.*, vol. 13, no. 9, pp. 4693–4707, Sep. 2014.
- [184] A. El Shafie, "Space-time coding for an energy harvesting cooperative secondary terminal," *IEEE Commun. Lett.*, vol. 18, no. 9, pp. 1571–1574, Sep. 2014.
- [185] X. Liu, F. Li, and Z. Na, "Optimal resource allocation in simultaneous cooperative spectrum sensing and energy harvesting for multichannel cognitive radio," *IEEE Access*, vol. 5, pp. 3801–3812, 2017.

- [186] F. Yao, H. Wu, Y. Chen, Y. Liu, and T. Liang, "Cluster-based collaborative spectrum sensing for energy harvesting cognitive wireless communication network," *IEEE Access*, vol. 5, pp. 9266–9276, 2017.
- [187] S. Yin, Z. Qu, Z. Wang, and L. Li, "Energy-efficient cooperation in cognitive wireless powered networks," *IEEE Commun. Lett.*, vol. 21, no. 1, pp. 128–131, Jan. 2017.
- [188] D. K. Verma, R. Y. Chang, and F.-T. Chien, "Energy-assisted decode-andforward for energy harvesting cooperative cognitive networks," *IEEE Trans. Cognit. Commun. Netw.*, vol. 3, no. 3, pp. 328–342, Sep. 2017.
- [189] C. Zhai, J. Liu, and L. Zheng, "Relay-based spectrum sharing with secondary users powered by wireless energy harvesting," *IEEE Trans. Commun.*, vol. 64, no. 5, pp. 1875–1887, May 2016.
- [190] J. Ren, Y. Zhang, N. Zhang, D. Zhang, and X. Shen, "Dynamic channel access to improve energy efficiency in cognitive radio sensor networks," *IEEE Trans. Wireless Commun.*, vol. 15, no. 5, pp. 3143–3156, May 2016.
- [191] S. D. Arthur Nkalango, H. Zhao, Y. Song, and T. Zhang, "Energy efficiency under double deck relay assistance on cluster cooperative spectrum sensing in hybrid spectrum sharing," *IEEE Access*, vol. 8, pp. 41298–41308, 2020.
- [192] S. Lee and R. Zhang, "Cognitive wireless powered network: Spectrum sharing models and throughput maximization," *IEEE Trans. Cognit. Commun. Netw.*, vol. 1, no. 3, pp. 335–346, Sep. 2015.
- [193] A. Jindal, A. Schaeffer-Filho, A. K. Marnerides, P. Smith, A. Mauthe, and L. Granville, "Tackling energy theft in smart grids through datadriven analysis," in *Proc. Int. Conf. Comput., Netw. Commun. (ICNC)*, Feb. 2020, pp. 410–414.
- [194] A. Jindal, G. S. Aujla, and N. Kumar, "SURVIVOR: A blockchain based edge-as-a-service framework for secure energy trading in SDN-enabled vehicle-to-grid environment," *Comput. Netw.*, vol. 153, pp. 36–48, Apr. 2019.
- [195] R. Chaudhary, A. Jindal, G. Aujla, S. Aggarwal, N. Kumar, and K. Choo, "Best: Blockchain-based secure energy trading in sdn-enabled intelligent transportation system," *Comput. Secur.*, vol. 85, pp. 288–289, Aug. 2019.
- [196] G. Praveen, V. Chamola, V. Hassija, and N. Kumar, "Blockchain for 5G: A prelude to future telecommunication," *IEEE Netw.*, vol. 34, no. 6, pp. 106–113, Nov./Dec. 2020.
- [197] P. Gorla, V. Chamola, V. Hassija, and D. Niyato, "Network slicing for 5G with UE state based allocation and blockchain approach," *IEEE Netw.*, early access, Nov. 17, 2020, doi: 10.1109/MNET.011.2000489.
- [198] P. Gorla, V. Chamola, V. Hassija, and N. Ansari, "Blockchain based framework for modeling and evaluating 5G spectrum sharing," *IEEE Netw.*, early access, Nov. 2, 2020, doi: 10.1109/MNET.011.2000469.
- [199] V. Hassija, V. Saxena, and V. Chamola, "A blockchain-based framework for drone-mounted base stations in tactile Internet environment," in *Proc. IEEE Conf. Comput. Commun. Workshops (INFOCOM WKSHPS)*, Jul. 2020, pp. 261–266.
- [200] A. Jindal, A. Marnerides, A. Scott, and D. Hutchison, "Identifying security challenges in renewable energy systems," in *Proc. 10th ACM Int. Conf. Future Energy Syst.*, 2019, pp. 370–372.



SANDEEP JOSHI (Senior Member, IEEE) received the B.Tech. degree in electronics and communication engineering from Uttar Pradesh Technical University, India, in 2006, the M.E. degree in communication engineering from the Birla Institute of Technology & Science Pilani, India, in 2009, and the Ph.D. degree in electrical engineering from the Indian Institute of Technology Delhi, in 2019.

From July 2008 to December 2008, he was an Interim Intern with Qualcomm India Pvt., Ltd., Hyderabad. From February 2009 to June 2009, he was a Senior Research Associate with Evalueserve Pvt., Ltd., Gurgaon. From July 2009 to June 2011, he was a Faculty Member of the Department of Electronics and Communication Engineering, Jaypee Institute of Information and Technology University, Noida. From July 2011 to December 2014, he was a Founding Faculty Member of the Department of Electronics and Communication Engineering, School of Engineering, Shiv Nadar University, Noida. From August 2019 to November 2019, he was an Early-Doctoral Research Fellow with the Department of Electrical Engineering, Indian Institute of Technology Delhi. From November 2019 to August 2020, he was a Chief Engineer with the Mobile Communications research group, Samsung R&D Institute India-Bangalore, Bengaluru. Since August 2020, he has been with the Birla Institute of Technology & Science Pilani, where he is currently an Assistant Professor with the Department of Electrical and Electronics Engineering. His research interests include communication theory, device-to-device communications, smart grid communications, molecular communications, and performance analysis of wireless communication systems.



SAINATH BITRAGUNTA (Senior Member, IEEE) received the B.Tech. degree in electronics and communications engineering from SVUCE, Tirupati, in 2001, the M.Tech degree from NIT Calicut, India, in 2007, and the Ph.D. degree in electrical communication engineering from the Indian Institute of Science (IISc), Bengaluru. After a brief stint at Zuari Cements as a Graduate Engineer Trainee, he joined and served as a faculty member of various academic institutions, from October 2001 to

November 2004. He worked at Reliance Communications, Navi Mumbai, from June 2007 to December 2008. He published peer-reviewed journal articles in reputed journals, like IEEE TRANSACTIONS ON COMMUNICATIONS, IEEE TRANSACTIONS ON WIRELESS COMMUNICATIONS, *IET Communications*, and *Physical Communication*. He also published in various IEEE conferences, out of which two papers won the best paper awards. He has more than ten years of teaching, research, and industry experience. He is currently working as an Assistant Professor with the Department of Electrical and Electronics Engineering, BITS Pilani, Pilani Campus. His research interests include communication systems analysis, design, and modeling. He is currently a member of the IEEE ComSoc and a Reviewer of IEEE TRANSACTIONS and IEEE SYSTEMS JOURNAL.



ASHISH PADHY received the B.Tech. degree in electronics and telecommunications engineering from the International Institute of Information Technology (IIIT), Bhubaneswar, India, in 2018. He is currently pursuing the M.E. degree in communications engineering from the Birla Institute of Technology & Science (BITS) Pilani, India. He has worked on projects related to biomedical signal processing and 5G communication systems. His research interests include designing efficient

wireless communication systems, applications of signal processing, machine learning, and deep learning in communication systems.



VINAY CHAMOLA (Senior Member, IEEE) received the B.E. degree in electrical and electronics engineering and the master's degree in communication engineering from the Birla Institute of Technology & Science Pilani, India, in 2010 and 2013, respectively, and the Ph.D. degree in electrical and computer engineering from the National University of Singapore, Singapore, in 2016. In 2015, he was a Visiting Researcher with the Autonomous Networks Research Group (ANRG),

University of Southern California, Los Angeles, CA, USA. He also worked as a Post-Doctoral Research Fellow with the National University of Singapore, Singapore, where he worked in the area of the Internet of Things. He is currently an Assistant Professor with the Department of Electrical and Electronics Engineering, BITS-Pilani, Pilani Campus, where he heads the Internet of Things Research Group/Lab. He has over 50 publications in high ranked SCI journals, including more than 33 IEEE TRANSACTIONS and journal articles. His works have been published in journals, like IEEE TRANSACTIONS ON COMMUNICATIONS, IEEE TRANSACTIONS ON VEHICULAR TECHNOLOGY, IEEE JOURNAL ON SELECTED AREAS IN COMMUNICATIONS, and IEEE Communications Magazine. His research interests include IoT security, blockchain, 5G network management, and addressing research issues in VANETs and UAV networks. Furthermore, his works have been accepted and presented in reputed conferences, like IEEE INFOCOM, IEEE GLOBECOM, IEEE ICC, and IEEE PerCom. He has served as a reviewer for several IEEE/Elsevier journals. He is a Guest Editor of Computer Communications (Elsevier). He serves as an Associate Editor for the Ad Hoc Networks (Elsevier), IET Quantum Communication, IET Networks, and Frontiers in Communications and Networks journal.



BIPLAB SIKDAR (Senior Member, IEEE) received the B.Tech. degree in electronics and communication engineering from North Eastern Hill University, Shillong, India, in 1996, the M.Tech. degree in electrical engineering from the Indian Institute of Technology Kanpur, Kanpur, India, in 1998, and the Ph.D. degree in electrical engineering from Rensselaer Polytechnic Institute, Troy, NY, USA, in 2001. He was a Faculty at Rensselaer Polytechnic Institute, from 2001 to

2013, first as an Assistant Professor and then as an Associate Professor. He is currently an Associate Professor with the Department of Electrical and Computer Engineering, National University of Singapore, Singapore. He serves as the Vice Dean for Graduate Program and as the Area Director for the Communications and Networks Lab, NUS. His current research interests include wireless networks, and security for Internet of Things and cyber physical systems. He is a member of Eta Kappa Nu and Tau Beta Pi. He is a Distinguished Speaker of ACM and a Fellow of the Institution of Engineers, Singapore. He has received many reputed fellowships, such as Leiv Eiriksson Fellowship (Norway), in 2011, and JSPS Fellowship (Japan), in 2010. He has served as a TPC member of various conferences, such as IEEE LANMAN, GLOBECOM, BROADNETS, and ICC. He served as an Associate Editor for the IEEE TRANSACTIONS ON COMMUNICATIONS, from 2007 to 2012. He currently serves as an Associate Editor for the IEEE TRANSACTIONS ON MOBILE COMPUTING. Additionally, he has served as a guest editor for various journals, including Future Generation Computing Systems (FGCS) (Elsevier), IEEE SENSORS JOURNAL, and Journal of Information Security and Applications (Elsevier).

...